

National Library of Canada Stationalism nationalism Consider

Canadian Theses Service

Service des thèces canadiennes

Course, Consde KIA 684

NOTICE

The quality of this microform is heavily dependent upon the quality of the original thesis submitted for microfilming. Every effort has been made to ensure the highest quality of reproduction possible.

If pages are missing, contact the university which granted the degree.

Some pages may have indistinct print especially if the eriginal pages were typed with a poor typewitter ristion or if the university cent us an interior photocopy.

Reproduction in full or in part of this microform is governed by the Canadian Copyright Act, R.S.C. 1970, c. C-30, and subsequent amendments.

AVIS

La qualité de cette microforme dépend grandement de la qualité de la thèse soumise au microfilmage. Neus avons tout fait pour assurer une qualité supérioure de reproduction.

S'Il manque des pages, vaullez communiquer avec l'université qui a conféré le grade.

La qualité d'impression de certaines pages peut laisser à désirer, aurtaut el les pages originales ent élé dadylographiées à l'aide d'un rubain usé du el l'université neus a let parvenir une photocopie de qualité injérieure.

La reproduction, même pertielle, de cette microforme est soumble à la Lei canadienne sur le droit d'auteur, SAC 1970, c. C-30, et ses amendements autosquerte.

The state of the s



UNIVERSITY OF ALBERTA

A COMPARATIVE STUDY OF PAPER AND PENCIL AND THE COMPUTER FOR PROCESS WRITING BY FOUR GRADE SIX WRITERS

BY

PAMELA ELLEN CARTER

A THESIS

SUBMITTED TO THE FACULTY OF GRADUATE STUDIES AND
RESEARCH IN PARTIAL FULFILLMENT OF THE REQUIREMENTS FOR
THE DEGREE OF MASTER OF EDUCATION

DEPARTMENT OF ELEMENTARY EDUCATION

EDMONTON, ALBERTA FALL, 1990



elanoitan eupérholidi@ du Canada

Canadian Theses Service Service des thèses canadiannes

The author has granted an irrevocable nonexclusive floence allowing the National Library of Canada to reproduce, loan, distribute or sell copies of his/her thesis by any means and in any form or format, making this thesis available to interested persons.

The author retains ownership of the copyright in higher thesis. Neither the thesis nor substantial extracts from it may be printed or otherwise reproduced without hie/her permission.

L'auteur a accordé une licence irrévocable et non exclusive permettant à la Bibliothèque nationale du Canada de reproduire, prêter. distribuer ou vendre des copies de sa thèse de quelque menière et sous quelque forme que ce soit pour mettre des exemplaires de cette thèse à la disposition des personnes intéreccées.

L'auteur conserve la propriété du droit d'auteur qui protège sa thèse. Ni la thèse ni des extrale substantiels de celle-ci ne doivent être imprimée ou autrement reproduits sens son autorisation.

ISBN 0-315-64899-6



UNIVERSITY OF ALBERTA

RELEASE FORM

NAME OF AUTHOR:

Pamela Ellen Carter

TITLE OF THESIS: A Comparative Study of Paper and Pencil and the Computer for

Process Writing by Four Grade

Six Writers

DEGREE: Master of Education

YEAR THIS DEGREE GRANTED: 1990

PERMISSION IS HEREBY GRANTED TO THE UNIVERSITY OF ALBERTA LIBRARY TO REPRODUCE SINGLE COPIES OF THIS THESIS AND TO LEND OR SELL SUCH COPIES FOR PRIVATE, SCHOLARLY OR SCIENTIFIC RESEARCH PURPOSES ONLY.

THE AUTHOR RESERVES OTHER PUBLICATION RIGHTS, AND NEITHER THE THESIS NOR EXTENSIVE EXTRACTS FROM IT MAY BE PRINTED OR OTHERWISE REPRODUCED WITHOUT THE AUTHOR'S WRITTEN PERMISSION.

> (Student's Signature) #9-10453-20 Que (Student's Permanent Address) Edmostow. alberta T6J 5AI

Date: Oct. 3. 1990

UNIVERSITY OF ALBERTA

FACULTY OF GRADUATE STUDIES AND RESEARCH

THE UNDERSIGNED CERTIFY THAT THEY HAVE READ, AND RECOMMEND TO THE FACULTY OF GRADUATE STUDIES AND RESEARCH FOR ACCEPTANCE, A THESIS ENTITLED "A COMPARATIVE STUDY OF PAPER AND PENCIL AND THE COMPUTER FOR PROCESS WRITING BY FOUR GRADE SIX WRITERS" SUBMITTED BY PAMELA ELLEN CARTER IN PARTIAL FULFILLMENT OF THE REQUIREMENTS FOR THE DEGREE OF MASTER OF EDUCATION.

M.P. Brown, Supervisor

D. Bayada

I E. Oster

Date: Kept. 14, 1990

ABSTRACT

This exploratory study compared the types of revisions done by grade six writers composing with paper and pencil and composing using a word processor at the computer. Completed compositions were rated to determine quality, and a word count was done to determine quantity of writing, when students used paper and pencil versus the computer for process writing.

The students, during process writing, composed on self-chosen topics in the natural setting of their classroom. They wrote, revised, conferenced with their teacher and classmates, and shared their writing.

Data consisted of all student compositions in both writing modes, video records of computer-generated stories, audio records of student-teacher conferences and of student and teacher interviews, and observations made by the researcher.

Analyses of the data were made by:

(1) categorizing and analyzing the revisions made throughout process writing, using a category system of revision developed for the study, having five categories: COMTENT, DEVELOPMENT, SENTENCE SENSE, CONVENTIONS, and STYLE (2) comparing the revisions independently initiated by the student with those that were teacher-initiated during conferencing (3) assessment of the final products using a four point holistic scale to determine and compare quality of

writing, and (4) a word count of all final products to compare quantity of writing in both modes of writing.

similar "vpes of revisions when composing with paper and pencil and with the computer. Most revisions were in the COMVENTIONS and SENTENCE SENSE categories. A greater number of revisions were made on the computer stories than on the paper-and-pencil stories. While conferencing with their teacher, a larger number of revisions were made on the computer stories were made on the computer stories. However, the students made more independent revisions on their paper-and-pencil stories. These grade six writers composed more at the computer and the stories were of better quality than those composed with paper and pencil.

Implications for teaching and research in the area of process writing are included in this report.

ACIDIONLEDGEMENTS

I would like to express my sincere thanks to those people who assisted me during the time of this study:

To Dr. Patti Browne, a special thanks for her encouragement, caring, and support throughout the study and during the final stages of the preparation of the report;

To Dr. D. Sawda and Dr. J. Oster for their valuable assistance and advice;

To the students who participated enthusiastically in the study, especially when they worked at the computers;

To the students' teacher who cooperated so willingly, and gave of her own time to help make my study a success;

To friends who encouraged me to completion of the final report, especially Connie, Ronnie, Wendy, and Barb, for whose constant support I was very grateful; and especially

To Mom, Sis, Ed, Pamela, and Shane for their continued support and confidence in all my endeavors.

TABLE OF CONTENTS

CHAPTER 1

INTRODU	CTION AND STATEMENT OF THE PROBLEM	1
	CHAPTER OVERVIEW	1
	PERSONAL REFLECTIONS	1
	BACKGROUND TO THE STUDY	2
	PURPOSE OF THE STUDY	4
	RESEARCH QUESTIONS	5
	DEFINITION OF TERMS	5
	LIMITATIONS OF THE STUDY	6
	SIGNIFICANCE OF THE STUDY	7
	OVERVIEW OF THE STUDY	8
CHAPTER 2		
REVIEW	OF THE RELATED LITERATURE	10
	CHAPTER OVERVIEW	10
	INTRODUCTION	10
	THE WRITING PROCESS	11
	THE COMPUTER AND THE WRITING PROCESS	12
	THE WRITING PROCESS AND REVISION	
	RESEARCH	15
	PAPER-AND-PENCIL STUDIES	15
	WORD PROCESSING STUDIES	22
	TOWARD A RESEARCH PARADIGH	30
	QUALITATIVE RESEARCH	31

REVISION CATEGORIES	32
QUALITY AND QUANTITY OF WRITING	33
SUMMARY	34
CEAPTER 3	
DESIGN AND PROCEDURES OF THE STUDY	36
CHAPTER OVERVIEW	36
INTRODUCTION	36
SELECTION OF THE SITE	37
SELECTION OF THE STUDY SETTING	37
DESCRIPTION OF THE STUDY CLASS AS A	
WRITING ENVIRONMENT	38
SELECTION OF THE WORD PROCESSING	
PROGRAM	39
SELECTION OF THE STUDY SAMPLE	40
DESIGN OF THE STUDY	42
DATA COLLECTION	44
DESCRIPTION OF THE TIME FRAME	44
COLLECTION OF THE WRITING SAMPLES	
USING PAPER AND PENCIL	45
COLLECTION OF THE WRITING SAMPLES	
USING THE COMPUTER	46
STUDENT INTERVIEWS	47
TEACHER INTERVIEW	48
COSERVATION	48
DATA AMALYSES PROCEDURES	48
RELIABILITY OF HOLISTIC SCORING	50

VERIFICATION OF REVISION CATEGORIES	52
CHAPTER 4	
FINDINGS OF THE STUDY	54
CHAPTER OVERVIEW	54
OVERVIEW OF THE APPLICATION OF THE	
REVISION CATEGORIES TO THE WRITING	
SAMPLES GATHERED IN THIS STUDY	55
CONTENT REVISIONS	55
DEVELOPMENT REVISIONS	56
SENTENCE SENSE REVISIONS	57
CONVENTIONS REVISIONS	58
STYLE REVISIONS	59
ANALYSES OF THE STUDENTS' REVISIONS BY	
CATEGORY	60
MICHELLE'S REVISIONS	61
MEGHAN'S REVISIONS	64
SARAH'S REVISIONS	65
JOAN'S REVISIONS	69
SUMMARY OF FINDINGS PERTAINING TO	
STUDENTS REVISIONS	71
A COMPARISON OF THE NUMBER OF	
INDEPENDENTLY INITIATED AND TEACHER-	
INITIATED REVISIONS	74
MICHELLE'S REVISIONS	74
NEGHAN'S REVISIONS	76
SARAH'S REVISIONS	76

JOAN'S REVISIONS	80
SUMMARY OF FINDINGS OF THE NUMBER OF	
INDEPENDENTLY INITIATED AND TEACHER-	
INITIATED REVISIONS	82
HOLISTIC SCORING OF STORIES	85
HOLISTIC SCORES OF MICHELLE'S	
STORIES	89
HOLISTIC SCORES OF MEGHAN'S	
STORIES	90
HOLISTIC SCORES OF SARAH'S	
STORIES	91
HOLISTIC SCORES OF JOAN'S	
STORIES	93

CONCLUDING STATEMENT	111
BIBLIOGRAPHY	113
APPENDIX 1. LETTER TO PARENTS	120
APPENDIX 2. STUDENT INTERVIEW DATA	122
APPENDIX 3. TEACHER INTERVIEW DATA	134
APPENDIX 4. HOLISTIC SCALE	137
APPENDIX 5. STUDENTS' STORIES	139

LIST OF TABLES

TABLE.	1	Revision Categories - Michelle's Paper- and-Pencil Stories	62
TABLE	2	Revision Categories - Michelle's Computer Stories	63
TABLE	3	Revision Categories - Meghan's Paper- and-Pencil Stories	64
TABLE	4	Revision Categories - Meghan's Computer Stories	66
TABLE	5	Revision Categories - Sarah' Paper-and- Pencil Stories	67
TABLE	6	Revision Categories - Sarah's Computer Stories	68
TABLE	7	Revision Categories - Joan's Paper-and- Pencil Stories	69
TABLE	8	Revision Categories - Joan's Computer	

TABLE 16	Independently Initiated and Teacher- Initiated Revisions in Joan's Computer Stories	82
TABLE 17	Percentage of Revisions in Both Composing Modes when Independently- Initiated and Teacher-Initiated	83
TABLE 18	Percentage of Revisions on Paper-and- Pencil Stories and Computer Composed Stories	85
TABLE 19	Holistic Scoring of Paper-and-Pencil Stories	87
TABLE 20	Holistic Scoring of Computer Stories	88
TABLE 21	Holistic Scoring of Michelle's Paper- and-Pencil Stories	89
TABLE 22	Holistic Scoring of Michelle's Computer Stories	90
TABLE 23	Holistic Scoring of Meghan's Paper-and- Pencil Stories	91
TABLE 24	Holistic Scoring of Meghan's Computer Stories	92
TABLE 25	Holistic Scoring of Sarah's Paper-and- Pencil Stories	93
TABLE 26	Holistic Scoring of Sarah's Computer Stories	94
TABLE 27	Holistic Scoring of Joan's Paper-and- Pencil Stories	95
TABLE 28	Holistic Scoring of Joan's Computer Stories	96
TABLE 29	Quality of Paper-and-Pencil and Computer Stories	97
TABLE 30	Length of Paper-and-Pencil and Computer Stories	98
TABLE 31	Quantity of Revisions in SENTENCE SENSE	102

CHAPTER 1

INTRODUCTION AND STATEMENT OF THE PROBLEM

Chapter Overview

The introductory chapter states some personal reflections and theoretical background to the study, the study's purpose, and the research questions which were the focus of the study. Limitations and significance are outlined and terms pertinent to the study are defined. Finally, an overview of the study and contents of each chapter are briefly described.

Personal Reflections

The arrival of the microcomputer made the decade of the eighties an exciting and invigorating experience for most students and many teachers. As a teacher, I had read articles about such things as LOGO (a one page article in MacLeans magazine which made little sense to me at the time, but somehow I knew it had significance for I tore it out and put it in a file folder) and how computers would change teaching. One could not help but wonder how this newest machine would impact education and teaching styles.

I had taken courses about microcomputers before I ever saw a microcomputer in a school. Soon I was

anxious to introduce my students to what I saw as the wonders of this newest piece of technological wizardry. With the support of a cautiously optimistic, but progressive principal, I set up the first two computers in our suburban Edmonton school. Recently, I reflected on the hours that it took that Sunday at school to read the manuals, set up and try the machine, and go back to the manuals, because the computer did not work! The cause for the reflection came almost seven years later when it took me twenty minutes to unpack, set up, and try the two new computers our school had just purchased. I looked around the classroom at the twenty or so computers that we now had and once again the excitement returned.

Thus, the interest for this study grew from those early experiences, and most importantly, from the delight and enthusiasm of the students I worked with both in the early and later years of my classroom computer experience.

Background to the Study

If, as Seymour Papert argued, the computer has the potential to be 'the pencil of the future' (Schwartz, 1982), we must begin now to study how it affects students as they write. Recent developments suggest that student composition should be considered as

process rather than product (Calkins, 1983; Graves, 1983; Harste, Woodward, and Burke, 1984) with the result that the writing paradigm has undergone a tremendous change and students in many classrooms are engaged in process writing. With the recursive nature of the writing process, we need to ask how microcomputer word processing programs can aid children in their development as writers (Branan, 1984; Kane 1983; and Schwartz, 1982).

The coupling of the word processor with process writing poses some new and interesting questions, in that there are many aspects of the writing program that could be studied. For example, students have long found the most difficult and tedious aspect of writing with paper and pencil to be the revision process (Daiute, 1985; Peacock, 1986). Students often resist the notion of revision, because they do not want to change their composition, especially where changes require that they cross out and erase what they have written (Calkins, 1980). If the word processor makes revision easier, the need exists to study its various effects, if any, on the revision process. Does the computer, as word processor, create an environment that will help students tackle the process of revision with a more positive attitude and with greater ease?

Many researchers and other educators argue that word processing is a catalyst for student growth in process writing. However, research in this area is still in its infancy. Many questions remain to be studied. Basically, we must continue to ask if ease of entering, revising, and producing attractive text are incentives for students to write more and to be more exhaustive in the revision process? Moreover, if true, will this lead to greater quantity and better quality writing by these young authors?

These questions were basic to the investigation undertaken and reported in this study.

Purpose of the Study

The major purpose of this study was to investigate the types of revisions done by student writers in grade six, when composing using a word processing program called <u>Magic Slate</u>, on an Apple IIe computer and an Imagewriter II printer. The revisions carried out at the computer were compared with the revisions done by the same students when composing with paper and pencil.

Completed compositions were also rated and compared to determine if there were any differences in quantity and quality of writing when students used the computer or paper and pencil as writing tools.

Research Questions

The following questions were the focus of the study:

- 1. What similarities and differences exist in the types of revisions done by grade six students when writing with paper and pencil as compared with writing using a word processor on the computer?
- 2. How does the conference, an integral component of process writing, affect the quantity of revisions done by writers at the computer and with paper and pencil?
- 3. Do students produce compositions of the same quality when using a word processing program and when using paper and pencil for composing?
- 4. Is there any difference in the quantity of writing produced when writing with paper and pencil compared with the amount produced when using a word processing program?

Definition of Terms

Mord Processing Program: a software program for a computer which allows the user to manipulate text (enter, move, erase, add, find, and replace), to format, and to print text. In this study the word

processing program will be <u>Magic Slate</u> (Sunburst Communications, 1984).

<u>Process Writing</u>: The recursive process of planning, writing, revising, editing, conferencing, sharing, and publishing written compositions.

Revision: any change made that will alter meaning or clarify an originally intended meaning; corrections in appearance, grammar, and spelling (Nold, 1982, p.97).

Revision Strategy: an idea or concern of the writer which helps explain why the writer changed something in his writing. The kind of revision may reflect the revision strategy used such as a cosmetic revision which would suggest a concern for appearance of the writing (Kopp, 1985, p.4)

Limitations of the Study

The following factors limit generalization of the findings:

- The <u>Magic Slate</u> word processing program and the Apple IIe computer each are only one of many word processing programs and computers available for students to use in the writing process.
- The sample size limits interpretations of the findings.

- 3. Though determined by their teacher to be competent in the use of the word processor, observation showed that the students needed to be more competent with the word processing program. Students who are more competent may produce better quality writing and this will need further research.
- 4. The investigator can not always observe the revision and editing strategies since some are done mentally, rather than verbally, on the computer or with a pencil.
- 5. The revision and editing process is recursive and time intensive. The study took place over two months and the time factor would limit the number writing samples that could be collected.

Significance of the Study

If the word processor is to be 'the pencil of the future', then a need exists to study its effects on student writing. Few studies have been done with elementary school students as subjects, though the microcomputer is being used in its word processing capacity in many elementary classrooms.

This study is intended to add to the body of research which is attempting to assess the effects of

word processors on the types of revisions done by young writers during the composing process.

Directives for the use of a word processor as a writer's tool may be determined from the conclusions reached from the study.

Overview of the Study

The report of this study is in five chapters.

Chapter 1 provides background information for the study; states the purpose, research questions, limitations, and significance of the study; defines terms; and outlines the contents of this report.

Chapter 2 reviews current literature on the writing process, the computer and the writing process, and writing and revision with paper and pencil and with the computer. The relevance, to this study, of the literature reviewed is the culmination of the chapter.

Chapter 3 presents the methodology used in the design of the study. The chapter discusses selection and description of the study site and sample, and describes procedures for data collection and data analysis.

Chapter 4 is the analysis of the data gathered from the sample, using both paper and pencil and a word processing program.

Chapter 5 reports the findings, draws conclusions, and states implications arising from the study.

CHAPTER 2

REVIEW OF THE RELATED LITERATURE

Chapter Overview

This chapter examines theories and research about the writing process. Particular attention is directed to those studies which have examined the revision process that is an important aspect of process writing.

Introduction

Concerns about improving writing have given rise to a broad range of literature, both theoretical and research based. Much of this literature is concerned with the improvement of writing abilities and how the school can teach children, and subsequently adults, to write better. While many approaches have been taken to solve the problem, the most recent approach has been to examine not just the final product, but the writing processes of children and adult writers. The intent is, of course, to provide us with insights and knowledge which can be used to improve our learning-to-write programs.

The Writing Process

Simply defined, writing is a <u>craft</u> (Graves, 1983). It is a process that involves thinking and communicating (Solomon, 1986). In writing, the author constructs meaning (Perl, 1979).

Prior to the early 1970's, writing was viewed as a linear process in which composing "consisted of pre-writing, writing and re-writing" (Faigley, Cherry, Jolliffe, Skinner, 1985, p.5). Emig (1971), Graves (1983), and Harste, Woodward, and Burke (1984) argued that writing was not as clear-cut a process as had once been believed. Solomon (1986) postulated that writing was an "awareness and understanding of the function and purpose of the whole. Children learn how to write by exploring the whole writing process. The parts by themselves are meaningless" (p.2).

Writing is a complex process. A major shift from the traditional paradigm is first attributed to the research of Emig (1971) who suggested that writing was non-linear and recursive. Three processes:

(1) planning, (2) writing, and (3) revision blend together. Perl (1979) and Sommers (1980) concurred that it is a recursive process. Peacock (1986) noted the interdependence of all aspects of process writing.

Flower and Hayes (1981) determined the linear aspect of the process modelled "the growth of the written product, not the inner process of the person producing it. Yet common sense and research tell us writers are constantly planning (prewriting), and revising (rewriting) as they compose (write), not in clear cut stages" (p.367).

The Computer and the Writing Process

Although cautiously optimistic about the merits of the computer as a tool for writing, researchers believe it is too early to make generalizations about its effects. Investigators report that writers enjoy using the computer to compose (Phenix and Hannan, 1984; Schwartz, 1982). Both quantity (Daiute, 1986; Fisher, 1983; Phenix and Hannan, 1984; Sekuler, 1985; Wheeler, 1985) and quality (Phenix and Hannan, 1984; Rodrigues, 1985; Sekuler, 1985) increased when young writers compose at the computer. Young writers have also been observed to do more risk taking in their writing (Phenix and Hannan, 1984; Schwartz, 1982), perhaps because the word processor does not have the permanence of the pencil, and changes can be made with ease.

Seymour Papert (1980) observed one student who moved from " . . . total rejection of writing to intense involvement accompanied by rapid improvement of

quality within a few weeks of beginning to write at the computer" (p.30). Yet all writers may not find that the computer fits into their style of writing and it may be more compatible to one part of the writing process than another (Ross, 1985).

Larter's (1987) study combined quantitative and qualitative data to investigate writing with and without the computer. The ten-month study involved 90 teachers and 180 students in grades 1, 3, and 6 from Toronto Board of Education Schools. It examined the writing process, teacher-pupil roles, teacher and pupil attitudes, teachers' philosophies about teaching writing, reading scores, and the products of writing. The experimental and control groups each had 15 teachers and 30 pupils. The students first wrote for 4 months with paper and pencil, followed by 6 months of writing using Commodore C64 computers. Pupils developed keyboarding skills before using the computer. The findings suggested that elementary students, most notably primary grades students, can write better with microcomputers than with paper and pencil, after a six month period. Larter stated the findings of the study demonstrated that some differences exist in the writing process itself when the two methods of composing were compared. This process also differs from grade to grade. Writers in grades one and three included a

picture with their paper-and-pencil stories, which was not always possible with the stories composed at the computer. The students wrote, read, and revised more when composing at the computer. Observational data indicated grade three students also had more on-task behaviour and were more motivated during the writing process. Grade six students did not read their writing to themselves as much as had been observed with grades one and three children.

Five eighth-grade students were investigated by Kane (1983) to determine how they used a microcomputerbased word processing system when composing. During ten sessions data were collected through (1) interviews with the students about their models for composing, experiences that had influenced their writing, and an evaluation of the computer as a writing medium (2) observation of students during writing class, and (3) hard computer copies of the students' work after each writing class. Kane concluded from the findings that students using the computer compose in the same general model and linear fashion that they had when writing with paper and pencil. Revisions were mainly the mechanics of writing (spelling, punctuation, grammar). The researcher also found evidence to suggest that the computer could be utilized in a positive way to promote involvement with composing, better organization of

text, and with the elimination of the drudgery of recopying, students also showed more interest in practicing revision procedures and experimenting with other forms of text.

The revision process emerges as the key in the discussion of process writing and the use of word processors. A writing area that Murray (1978) argued has been least researched, has taken on new prominence with the arrival of process writing and the word processor. It was the revision process that became the focus of this study. Revision processes are discussed in detail in the next section.

The Writing Process and Revision Research

Paper-and-Pencil Studies

Most young writers see writing as a linear process in which one draft with few revisions is produced. The first draft is often the only one (Faigley and Witte, 1981; Murray 1978). Students have long found the most difficult and tedious aspect of writing to be the revision process (Daiute, 1985; Peacock, 1986).

Research on revision has been scarce until most recently (Murray, 1978; Sommers, 1980), perhaps because it has been difficult to research with the traditional research techniques that have been used (Murray, 1978).

There have been few studies that have had revision as their exclusive research focus, with more studies including it as a variable in research on the composing process (Bridwell, 1980). Murray saw revision as a process of rewriting and argued that " . . . rewriting is one of the writing skills least researched, least examined, least understood, and-usually-least taught" (p.85).

From her studies of student writers, Calkins (1983) described revision as "audience-aware, reversible; . . . toying with options in the mind, trying things one way and then another." Calkins also saw revision as involving "shuttling back and forth between involvement and distance, between looking back and looking forward" (p.49).

Donald Murray (1978) discussed two forms of revision: internal and external. Internal revision is a process of discovery and development of meaning by the author during the time of the writing. The audience in this case is the writer. With external revision, the writer is concerned with a greater audience than self and revises in terms of the meaning the whole piece of writing will give to the audience.

In the last decade, views on revision have changed (Gentry, 1982; Monahan, 1984) with more emphasis on process, rather than product (Monahan, 1984) and the

acknowledgement that revision is a complex process
(Faigley and Witte, 1981; Gentry, 1982). Experienced
and inexperienced writers have different theories and
approaches to revision. Inexperienced writers are
concerned with fixing errors, while experienced
writers' concept of revision involves major changes
that may affect meaning and are more holistic (Faigley
and Witte, 1981; Murray, 1978; Sommers, 1980).
Revision, until recently, had been seen by many
teachers as merely editing and cleaning up of surface
mistakes, such as spelling and grammar (Faigley and
Witte, 1981). To students, revision has been viewed
more as punishment than as learning about their writing
(Emig, 1971; Murray, 1978).

The evidence from studies with good writers suggests that writing is not only linear, but also recursive (Faigley and Witte, 1981; Flower and Hayes, 1981; Murray 1978). The Flower and Hayes research, using verbal protocols, gave evidence to indicate that this recursiveness did indeed exist and that the composing process involved moving back and forth "among various activities of composing, and that expert writers frequently review what they have written and make changes while in the midst of generating text" (Faigley and Witte, 1981, p.400).

Donald Graves (1983), working with young writers, has shown that they are able to be far more sophisticated in their revision processes than had previously been believed. First revisions are usually simplistic, involving mechanical changes, Children move through a sequence in the types of revisions they do, until their revisions are at an advanced level involving changes in content and meaning.

The revision strategies of student writers and experienced adult writers were examined by Sommers (1980). Using a case study approach, twenty student writers, freshmen at Boston University and the University of Oklahoma, and twenty experienced adult writers from Boston and Oklahoma City were examined. Each subject wrote three drafts of three essays, totalling nine pieces of writing. The writers were interviewed following the completion of each essay. Each essay was analyzed and changes (revisions) were categorised by operation and levels of changes. The experienced and inexperienced writers not only revised differently, but their conceptualization of the process was different as well. The inexperienced writers saw revision as involving changes in words or phrases, frequently involving deletions. They had not developed strategies to see revision in terms of the whole essay. Revision was finished when the students felt "that they had not violated any of the rules for revising"

(p.383), the rules being based on conventions of spelling, grammar and format. Experienced writers defined revision differently. They revised to find a pattern, design or framework in their argument and to appeal to their readers. Sommers argued that this kind of revision is a process of discovering meaning together, in much the same way that Murray (1978) described rewriting. To experienced writers revision is a recursive process involving a variety of levels and cycles.

Bridwell (1980) assumed writing to be both a linear and a recursive process. In a study of 171 twelfth graders' transactional writing, the subjects wrote and revised one essay. A complex and exhaustive system of analyses was developed to evaluate the revisions made on the compositions. The investigator found that surface changes were made most often by the students. There existed patterns in revision strategies that students exhibited. The quality of their writing was not always an indication of the extent of their attempt to revise.

Having developed a taxonomy of revision that they described as simple, but robust, Faigley and Witte (1981) studied six inexperienced student writers, six experienced student writers and six expert adult

writers. The student writers were enrolled in a writing class at the University of Texas. The expert writers were professional writers in Austin, Texas; all had experience as journalists and three had published works of fiction. Over a three-day period the subjects composed, revised, and recopied their compositions. Using their taxonomy, which differentiated revisions that changed meaning in the writing and those that did not, the researchers evaluated the revisions made during each of the three drafts. Their findings supported Sommers' (1980), contention that experienced and inexperienced writers revise in different ways. The advanced writers made more changes in the macrostructure (meaning) of their essay, while the less experienced made more surface changes.

Monahan (1984) researched revision strategies with grade twelve students. The subjects were four basic (average) and four competent writers, as determined by a writing competency test. The eight subjects wrote three essays on predetermined topics and 'think aloud' protocols were used. Categories of revision were developed by Monahan, having studied similar categories developed by Bridwell, National Assessment of Educational Progress, and Sommers. Revisions were analysed in terms of the points of revision (stages from pre-writing to completion), levels of revision

(surface, clause, word, sentence, discourse), types of revision (addition, deletion, substitution, reordering, embedding) and purposes of each revision (cosmetic, mechanical, transitional, informational, stylistic).

The competent writers were found to revise for a peer audience, while the basic writers made more revisions for their teacher audience. Basic writers used the same revision strategies as competent writers, although they were more isolated and less frequently used.

Competent writers made more extensive revisions.

The National Association of Educational Progress (1977), in its second national assessment of writing, studied 2500 nine, thirteen, and seventeen year old students in the United States. Each group was given an assigned topic and subjects were permitted to revise and rewrite. A category system for revision was developed for the study. Using the category system for the analyses, it was determined that the nine year olds made more cosmetic changes while the older students made more holistic changes. Though all students made some revisions, the research showed that the 9 and 13 year old groups did not improve quality and organization, compared to the 17 year old group (p.9).

To date, most of the research published on writing and revision has focused on high school and college writers. In fact, much of the research centers on

advanced adult writers: senior college students,
English teachers, journalists (Faigley, Cherry,
Jolliffe, and Skinner, 1985). Krashen (1984), for
example, cites six studies, none of which had
elementary children as subjects. Summarizing from the
studies, he reported that for experienced writers,
revision was focused on content; for average writers
revision centered around changes to clarify meaning;
weak writers were concerned with mechanics.

The findings of high school and college studies suggest that we need to know more about the revision process, and in particular, the revision strategies used by writers in elementary grades (Gentry, 1982).

Word Processing Studies

Early research, dating from 1981, has provided conflicting data on the advantages and disadvantages of the word processor when writers revise their compositions. Hawrisher (1986a) emphasized that of the small body of knowledge that has emerged on writing and revision using the computer, it is difficult to draw comparisons owing to a wide variety of research designs, methods of data collection, variables examined, approach to data analyses, tools used, and competency levels of writers. Significant to these investigations is that early word processors were often

complex and unfamiliar to the writers. Hawrisher specifically points to <u>Bank Street Writer</u>, one of a very few word processors available for elementary students, in the early 1980's. Switching screens to write and then to revise or edit is not conducive to process writing and "emphasizes a linear rather than recursive process of composing" (Hawrisher, 1986a, p.12).

Several themes have consistently surfaced in the studies that have been done. Just as experienced and inexperienced writers compose differently, they also approach revision in a very different manner. Daiute (1985) stated that experienced writers have found they wrote and revised more on the computer, and the computer made their craft simpler. Collier (1983) concurred with Daiute, and added that the complex holistic tactics of revision used by experienced writers are very different from the simplistic revisions of the inexperienced - usually simple additions, substitutions or deletions.

The word processor is seen as having some advantages for young writers. The feature that appears most commendable, by both teachers and students, is that the word processor allows text to be revised and edited with comparative ease (Branan, 1984; Green, 1984; Schwartz, 1982). The opportunity to inset,

delete, move, search, and replace text, without having to rewrite the complete copy is seen to provide a more productive approach to the writing process. The ease of revising fosters a more positive attitude about writing (Rodrigues, 1985) and becomes a motivational factor for students to want to do more writing and revising (Bean, 1983; Collier, 1983; Daiute, 1982).

Students edit more frequently using a word processor than they do when using traditional methods, because it does not mean rewriting the whole piece, but simply making the needed changes (Daiute, 1986).

Negative attitudes toward revision develop from the 'drudgery of rewriting' and many students view rewriting as punishment for not correcting mistakes during the first draft (Balajithy, McKeveny and Lacitignda, 1986-87; Fisher, 1983; Schwartz, 1984).

Balajithy et al. further argued that "freed from the mechanical burden of recopying - the word processor can provide an environment where revision is both encouraged and easily accomplished" (p.28).

Smith (1985) argued the speed and simplicity of revising on the computer frees the writer to be more creative. The writer then experiments with language and the mental images the composer wants to capture are sustained. Daiute's (1983) interpretation is that with the rewriting and recopying removed from the writing

process, "many writers feel they are freer to be creative because mistakes have less physical consequence" (p.227).

Surveys and observations led researchers to postulate that anxiety about writing is decreased due to the ease of correction (Rodrigues, 1985).

Morocco (1986) described the word processor as having "features which create a powerful environment for acquiring revision strategies" (p.1). The large print screen text, continued Morocco, allowed children to reread and reflect on writing and helped to develop a good plan by being able to connect back to what they had written. It also facilitated group discussion in editing and revising, providing easier access for these activities to occur.

Collier (1983) noted that his "study of the effects of computer-based text-editors on the revising strategies of inexperienced writers is an attempt to begin mapping this largely unexplored territory" (p.149). The subjects were four female students, aged 19-32, enrolled in an introductory composition course, who could type, but had no word processing experience. Two of the subjects had average writing skills, one had weak skills, and the fourth had superior skills. The subjects wrote and revised at the computer once a week for six weeks. Two sessions were audio taped using

'think aloud' protocols. One session was video taped The writers' revision from the computer monitor. strategies and quality of writing was not significantly different when comparing computer written essays with paper and pencil writing. The average and superior writers benefited more from using the computer than did the weaker writer. Collier concluded that the results of the study did not suggest that the word processor was the answer to solving the long standing problems of revision, but that there did not appear to be detrimental effects on the strategies used in revising writing. Student revisions were not poorer when using the word processor and improved in those students that were good writers. Although the writing and revisions of the students remained unchanged, the revisions were done more extensively and more quickly using the word processor.

Hawisher (1986a) to determine the effects of word processing on revision strategies. The subjects wrote four essays alternately using pencil and computer. The three drafts for each essay were analyzed for revisions. An analytical scale was used to evaluate the quality of the writing. Findings indicated that the use of a word processor did not encourage students to make more revisions or increase the quality of

writing. The types of revisions were the same with both word processor and pen and typewriter .

Harris (1985) used a case study approach to investigate the word processor's effect on revising. This researcher believed that "rather extravagant and largely unsubstantiated claims" (p.323) were being made about word processors and their potential in improving composition writing by students. The subjects were six students in an honors freshman English course. The six subjects had both typing skills and computer experience. Students wrote four papers each and revised two with paper and pencil and two with the word processor. Findings of this study suggest that "word processing does not, in and of itself, encourage student writers to revise more extensively, especially in the macrostructure of a text" (p.330). Harris found the word processor to actually discourage revision.

Daiute's (1986) study of fifty-seven junior high school students in New York City investigated (1) revising patterns when students used paper and pencil and when they used the word processor (2) the effects of a word processor for revising as compared with a prompting program to guide students when revising compositions. Students were given time to acquire keyboarding skills and proficiency on the word processor to minimize the tool (the word processor) as

a major factor in the study. The students, aged 11-16, were average readers and good to poor writers. Their teacher was using the process approach to writing, having been trained on the model of Graves and Calkins The teacher had collaborated with the researcher for two years prior to the study, working out details of the study. Eight Apple II Plus microcomputers, two printers, a typing program, a word processing program, and a program to prompt revising were used for the study. Given assigned topics and a specific time limit, the nine-month study gathered writing samples at nine-week intervals. Writing was analysed using a taxonomy of revision adapted from Faigley and Witte's category system for revision. Conclusions of the study reported (1) these writers added more words to the end of text, but did not make more global revisions when they wrote using the computer as compared to using paper and pencil (2) students made more extensive revisions when a revision prompting program was added to the word processing program (3) the writing process is a complex one in which physical and cognitive processes interact (4) the instrument can affect the process in composition writing.

Research on the use of the word processor is in the early stages and there are few studies of young

writers using computers to revise their compositions. Schank (1986), using the <u>Bank Street Writer</u> word processing program, conducted a study of twenty-two grade four students to determine effects of writing on the computer versus traditional writing by hand. The students wrote on three assigned topics in a three week period. The control group used the traditional paper and pencil method, while the experimental group did all their work on the computer. Following student-teacher conferences on their stories the students rewrote and edited their work. The stories were analyzed in terms of editing and revision, using t-tests. Schank found no significant difference in the ways students revised and edited.

Kopp (1985) studied twelve grade five students in two Edmonton city schools. The students wrote one story each using the computer and one using paper and pencil. For the computer composed stories Apple II and Apple IIe microcomputers were used. The stories were written in three forty-minute periods. Data gathered through text revisions, questionnaires, personal observation, and interviews were analyzed for revisions, using the National Assessment of Educational Progress revision categories, and quality, using a holistic scoring scale. Kopp concluded that the sample revised more using the Bank Street Writer word

processing program than with paper and pencil and that the majority of revisions were those of mechanics of writing. Kopp also reported that his analysis showed the compositions composed using the word processor tended to be of better quality.

Toward a Research Paradigm

Evidence that studies needed to be done of young writers, writing with both paper and pencil and with computers, was quickly evident as I read the literature and research in this area. Most studies examined senior high school and college students, or adult writers (Bridwell, 1980; Collier, 1983; Faigley and Witte, 1981; Harris, 1985; Hawisher, 1986a; Monahan, 1984; Perl, 1979; Rodrigues, 1985; Sommers, 1980). Few studies of junior high school students (Daiute, 1985; De Almeida-van Hooydonk, 1986) and primary students (Graves, 1979; Larter, 1987; Phenix and Hannan, 1984) had been done. Studies that focused on students in grades four through six were also scarce (Daiute, 1985; Kopp, 1985; Larter, 1987; Schank, 1986) and of these, none had grade six students as the only subjects.

This need, and my interest, brought me to the study which compared grade six students writing with paper and pencil and with the computer, using a word processor.

Qualitative Research

The qualitative paradigm gained acceptance in the early seventies as a viable method of research, and the process of writing could more easily be studied. study of process writing was more conducive to qualitative rather than statistical methods. Hawisher's (1986b) survey isted only one ethnography. Some of the studies cited used some tools of ethnography: interview (Collier, 1983; Daiute, 1985; Hawisher, 1986a), video tapes (Collier, 1983; Daiute, 1985), and journals (Harris, 1985; Hawisher, 1986a; Rodrigues, 1985). Having considered the studies previously done, the topic I had chosen and the research questions that I had formulated, I determined that the study would be exploratory in nature and would also use several ethnographical tools: interviews, observations, videotapes, and audiotapes. These would be important in the analyses of the data. The instruments were then chosen that would best suit the purpose of answering the research questions. instruments are discussed in the next two sections of this chapter.

Revision Categories

In the studies cited, many of the researchers used revision categories as outlined by the National Assessment of Educational Progress Study (1977) or the categories of revision developed by Faigley and Witte (1981). Monahan (1984) developed his own categories of revision based on previous research and the work of Bridwell, Sommers and the National Assessment of Educational Progress. De Almeida van-Hooydonk (1986) developed a Category System of Revision that was "derived inductively from examination of all process drafts" (p.37).

The revision categories used in this study were drawn from the writing category system used by Edmonton Public Schools, as well as categories used in the school where the study was conducted.

CONTENT, DEVELOPMENT, SENTENCE SENSE, CONVENTIONS, and STYLE. Revisions in the CONTENT category were those which affected the central idea, the development of events, characters, or setting in the story. In the DEVELOPMENT category, revisions were those that affected the elements of development or the coherent sequence at the sentence or paragraph level of the story. In the SENTENCE SENSE category, revisions were

in the types and structures of the sentences, including additions and deletions of words and/or phrases.

Revisions in spelling, punctuation, capitalization, grammar, and paragraphing were classified as revisions in CONVENTIONS. Revisions in the STYLE category were those which affected the sensory images created and/or those which provide more precise words or details for the reader.

Quality and Quantity of Writing

Studies in which quality and/or quantity were variables were scarce. Hawisher (1986a) observed that revision strategies were not often studies in relation to quality of the writing. Hawisher used Diederich's analytical scale to analyse quality. Holistic scales were used by Bridwell (1980), Crawford (1989), Daiute (1986), and Kopp (1985). Daiute's and Bridwell's instrument was the scoring procedure developed by the Educational Testing Service in the United States. Kopp and Crawford used scales drawn from materials developed by Edmonton Public School Board and used in inservices with writing teachers. The holistic scale used in this study was also drawn from materials of the Edmonton Public School Board.

Daiute, 1986; Fisher, 1984; Phenix and Hannon, 1984; Sekuler, 1985; and Wheeler, 1985 state that

students write more when using the computer. Little data is yet available to support this claim. Daiute (1986) used words per text to determine length in her study. A decision was made to use a word count, as Daiute had, to determine length of the finished writing in this study.

Summary

Hawisher (1986b) reviewed studies that had been done on word processing. Only two of the twenty-four studies had elementary students in their sample. majority of the studies investigated college students and/or experienced writers. Methodology was case study, experimental (researcher manipulated the sample and assigned subjects randomly) or exploratory (subjects were not assigned randomly to groups). Only one ethnographical study was cited. In half of the studies, researchers concerned themselves with the question of the length of writing. A fewer number (seven of twenty four) did a holistic assessment of writing quality. Thirteen of the studies analysed revision, with a variety of tools being used, including student opinions, checklists, keystrokes, researcher criteria, and Faigley and Witte's categories of revision. In the studies that investigated attitudes about the use of the word processor, evidence existed

of a positive attitude. The variety of tools, writing tasks, variables, and methods of analyses indicate the paucity of information that is available about any one group of writers.

In most of the previous research, writers were given specific topics for their compositions. Graves (1983), Calkins (1983) and Harste et al. (1984) argued that students write best when they choose their own topics. This study investigated students in the composing process as it was carried on in the natural setting of their classroom and topics were student-chosen. The word processing program used for this study, Magic Slate, was the most recent and most technologically advanced for use by young writers, at the time of the study.

CHAPTER 3

DESIGN AND PROCEDURES OF THE STUDY

Chapter Overview

This chapter is a description of the study site, selection and description of the study sample and the word processing program, and procedures for data collection and data analyses.

Introduction

Graves (1983), Calkins (1983), and Harste,
Woodward and Burke (1984) argue that students write
best when they choose their own topics. Of the
research presently available, many studies took place
in an environment that did not represent the normal
classroom setting, with writing topics assigned by the
researcher or teacher. To provide a naturalistic
setting, this study investigated students in the
composing process as it was carried out in their
classroom under the direction of their teacher, with
self-chosen topics. Observation suggested that the
students soon became oblivious to the equipment.

Selection of the Site

Selection of the Study Setting

Having identified a need for a study centered in a naturalistic setting, in which students were actively involved in process writing, with self-chosen topics, a school was approached that might fit the criteria for the study.

Two classes, a grade 4/5 class and a grade 5/6 class, were observed on the first visit to the school. From the visit, it was determined that the school was involved in process writing, modelling Graves' philosophy of the writing process.

A decision was made to approach the teacher of the grade 5/6 class. At the time, I had not found any studies that had examined only grade six students and their revising processes. The grade six class had easiest access to daily use of the computers and had more experience using the computers. The study, its purpose, and its design were discussed with both the school administrator and the teacher that would be involved in the study. This teacher also believed the students were competent in the use of Magic Slate word processing programs, which was one of the word

Description of the Study Class as a Writing Environment

The study took place in the subjects' grade 5/6 classroom. It was a portable classroom, separate from the main section of the school, but connected by a hallway from the room that housed the computers. The students usually used the computers both in the computer room and in their own classroom. For the study, three computers were brought into the classroom each morning so that the subjects using the computers continued to be part of the regular classroom program.

Both through interview and observation, it was determined that the teacher had created a process writing environment that permeated each day. The students started their day with journal writing. The writing period was a daily period which lasted thirty to forty minutes. The students wrote for a variety of purposes and audiences. The majority of their writing was on self-chosen topics.

Each month the class produced a class newspaper which featured at least one article or story composed with paper and pencil and typed at the computer by each student in the class.

The students had opportunities to share their writing with individual classmates, the class as a whole, and their teacher. In addition, each Friday

afternoon the whole school came together to celebrate their writing successes and students from each grade shared their writing. Conferencing with their peers and with the teacher was a part of the composing process in this classroom.

Selection of the Word Processing Program

Magic Slate was the word processing software used for this study. This word processing software made use of icons, providing easily recognizable pictures for choice of the functions: edit text; load, save, print, or delete a file; start a new file; or make a new data disk.

The editing features were designed for ease of use, by even the youngest authors. Students could edit on the same screen as they had composed on, eliminating the need to change screens, which often proves difficult for young writers and interferes in the writing process.

A major feature of <u>Magic Slate</u> is its capability to allow writers to use 20-, 40-, or 80-column print. The 40-column print was used for the study. The students had been using the 40-column format for their work during the year. This also facilitated the data analyses, as the 40-column print was easily recorded on video record.

Selection of the Study Sample

Six grade six students, four girls and two boys, were selected from a class in a suburban Alberta school district. The students were in a split grade 5/6 class and were involved in daily writing as part of the total learning process.

The classroom teacher selected two good, two average, and two poor writers which she determined represented the writers in her grade six class. The teacher's decision was based on the evaluation of the writing done by her students during the preceding seven months of the school year. The students were also identified by their teacher as competent in the use of the word processor Magic Slate and were able to independently (1) load a file from a data disk (2) enter text (3) use the necessary text-editing features (delete, insert, move, add) (4) save text (5) print text.

Though the students were identified as competent users of the word processing program, they had not done any composing at the computer. They had previously used the word processor to type in stories that had been written with paper and pencil. The students had used the computer for many purposes throughout the year

and had good access to the computer room, as well as having frequent use of computers in their classroom.

I met with the students who had been selected by the classroom teacher. I explained that the study would take place during regular class time and that they would write some stories at the computer and some using their paper and pencil. I told them I was interested in what students did when they wrote stories. Students were not given any further information about the purposes of the study, as the researcher believed that it might affect how they wrote and revised their stories. All students selected agreed to participate in the study.

Written parental permission was received for each subject selected for the study. Students and parents were assured that a student could withdraw from the study at any time. The letter the parents received can be found in Appendix 1.

While six students participated in the study, the final sample consisted of four children and their writing samples. One student was dropped from the original six students identified by the teacher because of behavioral problems that affected performance.

Because the remaining five students consisted of four girls and one boy, a decision was made to drop the boy

so that gender issues would not have to be addressed in the analyses of the data.

To protect the anonymity of the students, the teacher, and the school in this study, they are referred to by the following pseudonyms: Michelle, Meghan, Sarah, Joan, Mrs. Taylor, and Hawthorne School.

Design of the Study

The subjects were divided into 2 groups of 3 students each. Group 1 wrote stories using the word processor for the first half of the study, while the second group wrote using pencil and paper. At the half way point in the study, the students changed modes, with the first group writing with paper and pencil and the second using the computer.

Three Apple IIe computer systems and one Apple
Imagewriter II printer were used for the computerassisted compositions. The 40-column version of Magic
Slate was used for the study. The students had been
using this version for their work during the year and
were familiar with it. Students saved their own
stories on individual data disks. A back-up disk was
used so that two copies of the students' work were
available at all times.

The students using paper and pencil were instructed to use pencil only on their paper for

writing, as had been done throughout the year in this class. I also instructed them to cross-out rather than erase any changes they made. This did not change the usual classroom procedure, as the teacher had encouraged them to cross-out, rather than erase, during their daily writing sessions.

Composing took place during the regular classroom writing program and was part of the authoring process that was being used in the classroom. Following each class, I made photocopies of the hand written work and hard copies of the computer-written stories.

Undoubtedly, the presence of the researcher and the technological equipment would provide some disruption to the normal classroom setting. The researcher and the teacher cooperated in keeping obvious distractions under control.

Throughout the study there were occasions on which I was not only a researcher, but a participant observer as well. All students in the class knew I was a teacher and would direct questions to me about their work, ask me to listen to their stories, or request help when they were having difficulty with part of the story. It was during these interactions that I often learned about the students' feelings and attitudes about writing.

Data Collectica

Data consisted of both paper-and-pencil and computer-produced student compositions, video-taped records of their computer compositions, audio records of their comments while composing under both conditions, student and teacher interviews, and researcher observations. These provided as complete a record as possible of the processes the students used when they composed and revised.

Description of the Time Frame

The study took place from April 9 through June 2, 1987. Thirty-one visits were made to the classroom. The students' writing class consisted of a 30 to 40 minute period each morning. In that time period, students wrote stories, conferenced with their teacher and their classmates, and shared their compositions with the class.

Prior to the commencement of the study, I visited the classroom on three occasions. The purpose was to familiarize myself with the classroom setting and the teacher's methods and to become acquainted with the students. Equipment that would be used in the study was set up. This gave the students, and particularly

those involved in the study, an opportunity to observe how the equipment would gather the data.

Collection of Writing Samples Using Paper and Pencil

The students composing with paper and pencil continued with their normal classroom procedures. The students were instructed to make all changes by crossing-out, rather than erasing. By using this method, all changes that the writers made were recorded on paper. This method was consistent with procedures usually used by students in their writing class.

At the end of each writing session, a photocopy of that day's work was made for my files. It was dated, so that all changes for that day's writing could be recorded.

An audio record of student and/or teacher comments was made while the students composed using paper and pencil. The audio recorder was placed on the desk next to the students' writing materials. Prior to the study the students had received an explanation of why I wanted them to use the audio recorder. As the study progressed I determined that this was difficult and disruptive for some of the students. While I encouraged them to try and make comments when they worked, I did not pressure the students. I matched any comments made by the students to the specific writing

revision, and co-ordinated comments made during conferences the students had with their teacher, with the copy of the student's writing done with the paper and pencil. This permitted me to verify changes in the composition and the reasons the students made these changes, where possible.

Collection of Writing Samples Using the Computer

As students composed using the word processor at the computer, each key stroke was recorded on video tape. Input from the processor was output through the video recorder (VCR) to the monitor, thus allowing a continual video recording. In this way, I was able to capture on video record, every change that the writer made while using the word processor.

At the end of each writing session, a hard copy of the day's work was made and the file saved for that date and version of the story (e.g. Humphrey.24.1). At the end of the next session, the file was again saved, but the date and version number were changed (e.g. Humphrey.25.2). This allowed me to keep a day-by-day record of all work, both on video record and on a hard copy. The hard copy could then be used to identify changes made, by comparing it with the video record.

While students composed at the computer an audio record was also made of student and/or teacher

comments. The audio recorder was placed on the desk next to the computer. As the study progressed I determined that this was difficult and disruptive for some of the students. I encouraged them to try and make comments when they worked, but did not pressure the students. I matched any comments made by the students to the specific writing revision, and co-ordinated comments made during conferences the students had with their teacher, with the copy of the student's writing done at the computer. This permitted the me to verify changes in the composition and the reasons the students made these changes, where possible.

Student Interviews

At the conclusion of the study, I had an audiorecorded interview with each student. The questions
presented were designed to gather data about the
students attitudes and perceptions about their writing
skills and their reactions to composing at the
computer.

This data, while not directly pertinent to the study, would provide background to the study. It could be important in interpreting the results of the study and drawing conclusions. Appendix 2 contains a summary

of the information gathered in the interviews with the students at the conclusion of the study.

Teacher Interview

An audio record of an interview with the teacher was also made. The teacher was interviewed to gather information about the subjects attitudes and abilities as emergent writers.

The teacher was also asked for information about the revision skills that she had been developing in her students, through process writing. Appendix 3 summarises the teacher's comments.

Observation |

I made notes during visits to the classroom on any aspect of the study that I felt may not have been recorded on video or audio tape.

Data Analyses Procedures

In keeping with the research questions, three areas were analyzed following the collection of the data. These were types of revisions by category, quality, and quantity of writing for all compositions composed under each writing condition. These analyses will be discussed in this section.

Each revision done by the students was recorded, categorised, and counted using the following revision categories:

CONTENT

Revisions which affect the central idea, the development of the events, characters, or setting in the story.

DEVELOPMENT

Revisions which affect the elements of development or the coherent sequence at the sentence or paragraph level of the story.

SENTENCE SENSE

Revisions in the types and structures of the sentences, including additions and deletions of words and/or phrases.

CONVENTIONS

Revisions in spelling, punctuation, capitalization, grammar, paragraphing.

STYLE

Revisions which affect the sensory images created and/or those which provide more precise words or details for the reader.

Revisions were divided into those made under each writing condition. The revisions were also divided into those done independently by the student and those done through teacher initiation. The types of revisions were compared and summarised for both students. In order not to skew the results of the study due to the limited keyboarding experience of the

study sample, where an error was obviously a keyboarding error and not a clear spelling error, it was not included in the error count in analyzing the data.

The quality of the compositions was analysed using a holistic scoring scale. Two raters scored each composition on a four-point scale. Scores of all paper-and-pencil compositions in the sample were compared with scores of stories written at the computer. The individual student's scores on paper-and pencil-stories were compared with the same student's computer-generated stories.

A number count was done on the final copy of each story in the sample to determine the quantity of the writing produced during the study. The study sample's volume on paper-and-pencil stories was compared to the volume produced by the same sample when using the computer to write stories.

Reliability of Melistic Scoring

Two raters were selected for assessing the student writers' compositions. These raters were teachers who had experience in a daily writing program and in assessing student composition at this grade level. A four-point holistic scoring scale (Appendix 4), used by

the Edmonton Public School Board (1986), was the tool used to evaluate the quality of the writings.

For consistency in scoring, the two raters and the researcher met for a four-hour session. The holistic scoring scale was discussed in detail by the raters and the researcher. The raters then took grade six students' writing samples and rated them independently. The raters discussed the scores assigned to the sample compositions and where discrepancies existed, they determined why the scores had differed. Both raters indicated that they felt the meeting had been extremely beneficial in helping them to score the writing samples gathered during the study.

The raters scored the writing samples from the study. In order to reduce bias, the researcher reproduced all the stories using a word processing program. As a result, raters did not know which compositions were computer generated and which were produced using paper and pencil. Copies of the selections in the format presented to the raters are in Appendix 5.

The Arrington Formula used by Kopp (1985) to determine percentage of agreement between the two raters was then applied to the scored compositions.

The percentage of agreement formula is:

2 X # of Agreed Scores

2 X # Agreed Scores + # Disagreed Scores

The percentage of agreement on the holistic scoring of the students' writing was 0.80.

Verification of Revision Categories

The revision categories that had been developed by the researcher were discussed with two language arts teachers.

The researcher had a two-hour meeting with the first teacher. The categories were discussed in detail. The teacher believed the revision categories to reflect the types of revisions sixth grade students would make in process writing. Together, and independently, the researcher and the teacher classified revisions from samples of compositions at this grade level.

The second language arts teacher met with the researcher for four hours. A similar procedure was adopted as with the first teacher. The compositions that the second teacher and researcher independently classified by revision categories were compared for percentage of agreement using the Arrington Formula.

The percentage of agreement on the revision categories was 0.90.

CEAPTER 4

FINDINGS OF THE STUDY

Chapter Overview

This chapter reports the analysis of the revisions done according to the revision categories. Examples of each revision category are presented.

The second section of this chapter presents the analyses of the revisions according to the revision categories designed for the study. Since the writing samples, except for one, involved conferencing with the classroom teacher, both independently initiated and teacher-initiated revisions are studied.

In the third section, the numbers of independently initiated and teacher-initiated revisions are reported. The final two sections of this chapter include the results of the holistic scoring of the paper-and-pencil and computer stories by two independent raters, as well as a comparison of the quantity of writing for each subject in paper-and-pencil and computer mode.

Overview of the Application of the Royleting Categories to the Writing Samples Cathered in this Study

The revision categories developed for the study and described in Chapter 3 (page 49) were used to identify changes made by the student writeks during process writing. Since these categories were specific to the study, this section is included to explain how the category system applied in this study by providing examples drawn from the writing samples of the children who participated. For each category, a comment on the overall percentage use of the specific Category in the writing samples produced during the study is also reported. The analysis of the subjects' individual writing products follows.

CONTENT Revisions

Revisions in CONTENT by all four students were less than 1% of the total revisions. Nost revisions in CONTENT were changes in the title of the story. The writer would assign a title and begin to write a story, only to find after a short time that either she could not think of anything else to write about that topic or that she thought of new ideas to fit with what she had written, but the new ideas did not match the title. These title changes were classified as CONTENT changes

because they affected the central idea and the development of the story.

In her story "Return of the Cubs", Sarah wrote and revised as follows:

ORIGINAL: Then one day the cubs were playing by a den of bears. What will happen? Will they survive? Find out in part three.

REVISED: Then one day the cubs were playing by a tent where people were. What will happen? Will they survive? Find out in part three.

The revision changed the central idea as well as the setting of what Sarah would write about in part three. The events would develop differently with the change in CONTENT. Sarah further revised this part of her story, with the final copy being:

REVISED: Then one day the cubs were playing by a campaite where people were camping. What will happen?

Will they run quickly to their parents?

Find out in part three.

DEVELOPMENT Revisions

The revisions that were classified under the category, DEVELOPMENT, were also small in number, just 1% of the total revisions. Most changes were those that involved the concept of time, so that the story would have a logical sequence.

During Michelle's conference about the story
"Redheed" an example of the change in time frame was
pointed out by her teacher. The teacher suggested that

perhaps someone would notice if a student came to class and then was gone for about two hours. Michelle then made the change, having the character in the story out of the room for about 45 minutes.

ORIGINAL: One day when Suzy went to school and when she got there everybody started laughing and teasing her for no good reason at all. She ran into the washroom and started to cry.

Later on at about 11 am, Sury went back into her classroom.

REVISED: One day when Susy went to school and when she got there everybody started laughing and teasing her for no good reason at all. She ran into the washroom and started to cry.

At about 9:45 A.M., Suzy went back into her classroom.

Neghan added a sentence to her story, "Toby the Turtle". It was categorised as a revision in the DEVELOPMENT category. This addition affected the development of the paragraph, but did not affect the central idea of the story.

ORIGINAL: Luke and his mother went outside to look for Tobby. "I found him!!" yelled Luke after they had been looking for awhile. He took him inside and gave him some food.

REVISED: Luke and his mother went outside to look for Toby. "I found him!!" yelled Luke after they had been looking for awhile. He must have been hiding under the shed. He took him inside and gave him some food.

SENTENCE SENSE Revisions

In the SENTENCE SENSE category writers added or deleted words and phrases at the sentence level, or

changed the structure of the sentence to better reflect what they wanted to say in their writing. Revisions under the SENTENCE SENSE category accounted for 37% of the changes the students made.

Joan made a number of revisions in the SENTENCE SENSE category. Examples from her story, "How Cats Eyes Glow" are:

ORIGINAL: One night the two cats didn't have to stare at the moon because their eyes started to glow and lit the path they were walking on.

REVISED: One night the two cats didn't need to stare at the moon because their eyes started to glow and lit the path they were crawling on.

ORIGINAL: But he needed some help from his two friends Freddy and Sam. So he called up his two friends to come over and help him tell the children the stories. Ten minutes later Sam and Freddy were over where all the children and Manaboso were sitting.

REVISED: But he needed some help from his two friends Freddy and Sam. So he phoned his two friends to come over and help him tell the children the story. Ten minutes later Sam and Freddy arrived and went over to where all the children and Wanabozo were sitting.

CONVENTIONS Revisions

In interviews with both the students and their teacher it was apparent that these writers considered revising to be those changes that come under the CONVENTIONS category. More than half the total

revisions (59%) done by the writers were in the CONVENTIONS category.

In Meghan's story, "Mystery of the Mouse Eyes,
Part 2", she made many revisions in the CONVENTIONS
category. Revisions were both independently initiated
and teacher-initiated during conferencing.

ORIGINAL: After Grandma went to bed mom asked me a few questions. Hony, about how tall was the men? "I din't know, about as tall as you." I answered. "Did you see the hair color?" she asked again.

REVISED: After Grandma went to bed mom asked me a few questions. Honey, about how tall were the men? "I don't know, about as tall as you." I answered. "Did you see the hair color?" she asked again.

ORIGINAL: "Oh mom, what's the matter?" asked Mary Jane.
"It's Uncle Jim, you were right, he is dead!"

REVISED: "Oh Mom, what's the matter?" asked Mary Jane.
"It's Uncle Jim. You were right. He is dead!"

STYLE Revisions

As with the revision categories of CONTENT and DEVELOPMENT, a very small amount, (2%) of the total revisions, were in the STYLE category.

Sarah, in her story, "The Problem", made a revision categorized under STYLE.

ORIGINAL: Tiny started to walk home...He replied harshly, I don't need a dome friend like you.

REVISED: Tiny started to walk home...He screamed harshly, I don't need a dome friend like you.

Michelle's story, "Mary Finds a Friend", also had a revision in the STYLE category.

ORIGINAL: Mary's brother's awed at the bear cub.

REVISED: Mary's brother's gazed in amazement at the bear cub.

The revisions made by Sarah and Michelle in the STYLE category gave their audience a better visual image and provided more precise words than the original writing.

Analyses of the Subjects' Revisions by Category

The first research question asked what similarities and differences exist in in the types of revisions done by grade six students when writing with paper and pencil as compared with using the word processor on the computer. In order to address this research question, the revisions made to every paper-and-pencil and computer story written by each of the four subjects (Michelle, Meghan, Sarah, and Joan) were classified according to the revision categories.

The revisions were also identified as independently initiated or teacher-initiated revisions in each category.

Michelle's Revisions

Michelle produced the greatest number of stories, three paper-and-pencil stories and six computer stories. According to her teacher, Michelle did not care to revise very much and was more anxious to write about the next idea she had in her mind.

On her paper-and-pencil stories (TABLE 1) the largest percentage of her revisions (63%) were in the SENTENCE SENSE category. She also made a significant number of revisions in the CONVENTIONS category (32%). Fewer numbers of revisions were made in the CONTENT category (2%) and the STYLE category (5%). Michelle made no revisions in the DEVELOPMENT category.

When independently initiated and teacher-initiated revisions are examined, the same pattern exists.

However, in the teacher-initiated revisions, the revisions were only in the SENTENCE SENSE category (91%) and CONVENTIONS category (9%).

Michelle exhibited a similar pattern in the revisions on her computer stories (TABLE 2). Most revisions were done in the SENTENCE SENSE category (50%) and the CONVENTIONS category (36%). A small

number of revisions were made in the other categories:
CONTENT (3%), DEVELOPMENT (5%), and STYLE (5%). The
number of teacher-initiated revisions in the SENTENCE
SENSE category (54%) was not as disproportionate as it
had been in the paper-and-pencil stories.

Michelle was identified by her teacher as a good student and writer. Her writing did not exhibit a spelling problem which affects the CONVENTIONS category for many writers.

TABLE 1

Revision Categories - Michelle's Paper-and-Pencil
Stories

<u> </u>						
55053	a		MARCH -	\$60/250/CS \$60/68	COMPANIENCE	9884A
Candy Non	*I	-	•	12 2	5 2	1
The Fountain	I	1	•	9 21	16 1	1
The Time of of Dinosaure		1	-	11 7	7 -	1 -
100721009- 10007000007		34	•	494	434	54
MALENTAGES SENCENS- SENCENS-		•	•	914	*	-
SOUT SEASTONS		*	•	639	321	54
• I - Indep		ntly I	itiated			

^{*} T - Teacher-Initiated

TABLE 2 Revision Categories - Michelle's Computer Stories

Front	œ	io de la constantina	PRINTS -	8000000 80008	COMPARIONS	877.65
Mary Finds A Friend	*I *T	1	:	-	5 13	1
Fun	I	1	ī	5 4	2 3	ī
Sick	I	ī	-	i	1	ī
The Big Adventure	I	1	-	9 12	6	i
Redheed	I	-	3	3 10	2 2	1
Allen	Ī	-	2 1	5	3	:
MAXISSONS-		64	44	444	421	44
MEYESTONS- TENCHESS- THETTENSED		19	•	549	334	9
MENISIONS SOUT		34	54	504	361	54
* I - Indep	ender er-Is	tly I	nitiated ed			

Meghan's Revisions

When writing with paper and pencil (TABLE 3)

Meghan made no revisions in the CONTENT or STYLE

categories. One revision was made in the DEVELOPMENT

category when initiated by the teacher in conferencing.

Meghan made most revisions in the CONVENTIONS category

(66%), followed by the SENTENCE SENSE category (33%).

When independently initiated and teacher—initiated

revisions are examined, more revisions were made in the

CONVENTIONS category when they were

TABLE 3

Revision Cotogories - Meghan's Paper-and-Pencil Stories

Story	α	arter!	DEVELOR-	SENTENCE SENIOR	CONVENTIONS	51114			
Earth Morm City	*I *T	-	ī	13 7	3 16	:			
Bebysitting	I	-	-	4 2	16 17	-			
11001-010-015 1001-010-015		-	•	479	534	•			
SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLSTONS- SMALLS		-	24	211	774	•			
2007 2007		-	19	339	661	•			
• I - Indep	• I - Independently Initiated • T - Teacher-Initiated								

independently initiated (53%) and teacher-initiated (77%) than in the SENTENCE SENSE category for both conditions (47% and 21%). The high percentage of revisions that were teacher-initiated in the CONVENTIONS category were boosted by the number of spelling and punctuation revisions that the teacher initiated. Though an average writer, Meghan was weak in these areas.

With the stories composed on the computer (TABLE 4), Neghan made no revisions in the CONTENT category, as was the case with her paper-and-pencil stories. Similarly, small portions of the revisions on her computer stories were in the DEVELOPMENT category (0.5%) and STYLE category (0.5%). With the computer stories 20% of the revisions were in the SENTENCE SENSE category and 79% were in the CONVENTIONS category. A pattern fairly similar to the paper-and-pencil stories was exhibited, with a high proportion of the teacher-initiated revisions being in the CONVENTIONS category (87%).

Sarah's Revisions

As with Michelle and Neghan, Sarah made the majority of her revisions under both writing conditions, in the SENTENCE SENSE and CONVENTIONS

categories. However some differences did exist in the pattern exhibited by Sarah.

With her paper-and-pencil stories (TABLE 5), the number of revisions in the CONTENT category (4%), the DEVELOPMENT category (4%) and the STYLE category (7%) were minimal. Sarah made 45% of her revisions in the SENTENCE SENSE category and 40% in the CONVENTIONS category.

TABLE 4

Revision Categories - Meghan's Computer Stories

55.05X	CO.	18 E S	PEASTON-	5000 CE	CUNERCUS	553.65
Rosebud	*I	•	-	6 2	10	•
Toby the Turtle	I	•	- 1	7 2	3 21	1
Mystery of the Mouse Eyes	I	-	-	12 6	45 54	ī
NEVZ STORE-	•	•	•	301	694	19
MALESCON- SENCENT- THE STATES		•	19	119	87%	10
MANISTONS BORNT		•	0.50	204	794	0.54
* I - Indep	pende ter-I	atly :	Initiated ted			

TABLE 5 Revisions Categories - Sarah's Paper-and-Pencil Stories

Stems (CONTRACTOR	DEVELOP- NESTE	8200.00CS 800008	CONVENEZORS	577.4
How the Earth Got its Color		ī	18 2	3 13	-
Return of * the Cubs *		3 -	18 4	7 14	3 3
REVISIONS- INDEPENDENT	29	5.54	681	194	5.54
revisions- Teacher- Initiated	78	31	154	601	76
rotal Revisions	41	41	454	401	76
* I - Independ * T - Teacher-					

When the independently initiated and the teacher-initiated revisions were examined, 68% of Sarah's independently initiated revisions were in the SENTENCE SENSE category and 19% were in the CONVENTIONS category. With teacher-initiated revisions, the pattern was reversed, with only 15% of the revisions being in the SENTENCE SENSE category and 68% of the revisions were in the CONVENTIONS category.

As with the paper-and-pencil stories, the stories that Sarah wrote on the computer (TABLE 6) had small

percentages of revisions in the CONTENT category (1%),
DEVELOPMENT category (1%) and STYLE category (1%). The
largest proportion of revisions were in the CONVENTIONS
category (71%) and the SENTENCE SENSE category (26%).
This was true for both independently initiated and
teacher-initiated revisions.

TABLE 6

Revision Categories - Sarah's Computer Stories

FFCM.Y	CO	SEA.S	DEVELOR-	SENTENCE SENTER	CONVENSIONS	57725
The Problem	*I *T	1	-	11 3	12 36	1
The Alley Cat	I	-	- 1	6 11	15 31	-
Rusty	I	-	-	2 -	8 6	-
The Clue	I	1	1	15 8	25 21	ī
REVISIONS- INDEPENDENT	ı	24	19	354	611	10
Mentricus- The Carry- The Carry-		•	29	104	794	19
SOEVT SOEVT		19	19	261	719	10
* I - Indep * ? - Teach	ende er-I	ntly :	Initiated ted			

Joan's Revisions

Joan's paper-and-pencil stories (TABLE 7) showed few revisions in the CONTENT category (1%), the DEVELOPMENT category (1%), and the STYLE category (3%), which followed the pattern exhibited by the other three student writers. The majority of the revisions were in the SENTENCE SENSE category (67%), followed by the CONVENTIONS category (28%). When the independently initiated and teacher-initiated revisions were examined

TABLE 7 Revision Categories - Joan's Paper-and-Pencil Stories

SECRIT .	- ai	louis -	PRINTS -		COMMERCE (COS	57714	-
Now Cats Eyes Glow	*I	1	1	45	20	3	
Untitled Story	Ī	-	ī	31 25	7 15	1 -	
11005-010015 11005-010015		10	16	704	254	34	
NEVESTONS- SENCERO- INTERACED		•	24	611	374	-	
SOEVT BEATE SOOKS		10	19	679	281	34	
· I - Inden	ander	tly I	nitiated				

^{- 1 -} Independently Initiated

• T - Teacher-Initiated

separately, the same pattern was evident for these two categories. Joan made no teacher-initiated revisions in the CONTENT and STYLE CATEGORIES when composing with paper and pencil.

Joan wrote two stories with paper and pencil. She chose to only conference one of those stories with her teacher. She took a long time when composing her first story with paper and pencil and made revisions each time she worked on the story. She did conference with her teacher on the second paper-and-pencil story.

Mhen composing at the computer (TABLE 8), Joan made no revisions in the CONTENT category and few in the DEVELOPMENT (0.5%) and STYLE (0.5%) categories.

None of the revisions in CONTENT, DEVELOPMENT or STYLE were initiated by the teacher on Joan's computer-composed stories. Of the total revisions done by Joan on her computer stories, the largest proportion were in the CONVENTIONS category (65%), followed by the SENTENCE SENSE category (34%). When these categories are further examined, Joan made an equal number of revisions in both the CONVENTIONS and the SENTENCE SENSE categories (49% in each category). The teacher-initiated revisions were only 12% in the SENTENCE SENSE category. The largest proportion were in the CONVENTIONS category (88%).

TABLE 8

Revision Categories - Joan's Computer Stories

95057	C	ra w as	HENTE OF THE PERSON NAMED IN COLUMN TWO IS NOT THE PERSON NAMED IN COLUMN TO THE PERSON NAMED IN		CONVENEZCOS	51114
Teenagers Knees	•I •T	-	-	15 6	13 8	- -
Goodbye Swee Valley High	tI T	-	1 -	22	24 37	1
REVISIONS- INCOPERCENT		-	19	494	494	19
revisions- Tracher- Initiated		-	-	129	881	•
MANISIONS LOSAT		-	0.54	341	651	0.54
• I - Indepe						

^{*} T - Teacher-Initiated

Summary of Findings Pertaining to Students' Revisions

The analysis of the writing revisions, according to the revision categories used, did reveal some similarities and differences when the paper-and-pencil and computer stories were compared. In both modes, neither CONTENT, DEVELOPMENT, nor STYLE revisions, which could be considered the higher order of revisions, exceeded 7% of the total number of revisions recorded. Three of the four students did not make revisions in one or more of these three categories in

their writing, regardless of mode, even when there was some teacher conferencing involved.

Revisions in the CONVENTIONS and SENTENCE SENSE categories accounted for the majority of changes both on paper-and-pencil and computer composed stories.

Each student exhibited a unique pattern.

Michelle had a higher percentage of SENTENCE SENSE revisions in both modes of writing, and less in the CONVENTIONS category. This pattern also held when independently initiated and teacher-initiated revisions were examined.

Meghan's revision patterns were reversed from those exhibited by Michelle, with over 65% of Meghan's revisions in both writing modes being changes in the CONVENTIONS category and a much smaller percentage being revisions in the SENTENCE SENSE category. This pattern also held when independently initiated and teacher-initiated revisions were examined.

Sarah made more revisions in the SENTENCE SENSE category than the CONVENTIONS category on her paper-and-pencil stories. The reverse pattern was exhibited on computer stories with the greater number of revisions in the CONVENTIONS category and considerably less revisions in the SENTENCE SENSE category. In the stories that Sarah composed with paper and pencil, she independently initiated more revisions in the SENTENCE

SENSE category than in the CONVENTIONS category. When the revisions were teacher-initiated, there were a much greater number in the CONVENTIONS category than in the SENTENCE SENSE category. A more consistent pattern was exhibited in Sarah's computer-composed stories with the revisions in the CONVENTIONS category, both independently initiated and teacher-initiated, far outnumbering those in the SENTENCE SENSE category.

Similar to Sarah, Joan's paper-and-pencil revisions exhibited a different pattern than those of her computer-composed compositions. With her paperand-pencil compositions, SENTENCE SENSE category revisions, both independently initiated and teacherinitiated, were greater than those in the CONVENTIONS category. In Joan's stories composed at the computer, the total revisions showed almost twice as many revisions in the CONVENTIONS category than in the SENTENCE SENSE category. When the independently initiated revisions were examined there were an equal number of revisions in both categories (49%). An examination of the teacher-initiated revisions revealed that the revisions in CONVENTIONS category were 88% of the total revisions and those in the SENTENCE SENSE category were only 12% of the total.

A Comparison of the Number of Independently Initiated and Teacher-Initiated Revisions

The second research question asked how the conference, an integral component of process writing, affected the quantity of revisions done by developing writers. In order to address the second research question, the number of revisions made on paper-and-pencil stories and computer stories were compared. The revisions made independently by the writers and those that were teacher-initiated through conferencing were compared.

During the coding of the revisions, a number count was also made of the independently initiated revisions and of the teacher-initiated revisions made during the conference.

Michelle's Revisions

Sixty-six percent of Michelle's revisions on her paper-and-pencil stories were independently initiated, and 34% of her revisions were teacher-initiated (TABLE 9).

On her computer-composed stories the reverse was evident. Thirty-eight percent of Michelle's revisions were independently initiated and 62% were teacher-initiated (TABLE 10). Michelle enjoyed working at the

computer and produced six stories during the study.

The stories were short and Michelle was observed to be very anxious to start new stories. This was reflected in the quantity of independently initiated revisions.

Independently Initiated and Teacher-Initiated Revisions
in Nichelle's Paper-and-Pencil Stories

Student	Teacher	Total
18	4	22
27	22	49
20	7	27
65	33	98
661	34%	
	18 27 20 65	18 4 27 22 20 7 65 33

Independently Initiated and Teacher-Initiated
Revisions in Michelle's Computer Stories

	Initia	ted by:	Total
Story	Student	Teacher	
Mary Finds a Friend	7	18	25
Fun	8	9	17
Sick	1	10	11
The Big Adventure	18	19	37
Redhead	6	16	22
Allen	10	10	20
Total revisions	50	82	132
% of revisions	384	624	

Meghan's Revisions

Forty-six percent of Meghan's paper-and-pencil revisions were independently initiated and 54% of her revisions were teacher-initiated (TABLE 11).

TABLE 11

Independently Initiated and Teacher-Initiated

Revisions in Neghan's Paper-and-Pencil Stories

Initia		
Student	Teacher	Total
16	24	40
20	19	39
36	43	79
464	54%	
	16 20 36	16 24 20 19 36 43

On stories composed at the computer, her number of revisions were very similar, with 47% being independently initiated and 53% being teacher-initiated (TABLE 12). In both modes, a small amount more of the revisions were teacher-initiated than were independently initiated. Meghan was identified by her teacher as an average writer, but had some difficulties with spelling and grammar, which may have accounted for the larger percentage of revisions being teacher-initiated.

TABLE 12

Independently Initiated and Teacher-Initiated

Revisions in Neghan's Computer Stories

	Initia			
Story	Student	Teacher	Total	
Rosebud	16	10	26	
Toby the Turtle	11	24	35	
Mystery of the Mouse	57	61	118	
Eyes				
Total revisions	84	95	179	
% of revisions	478	534		

Sarah's Revisions

independently initiated and 43% were teacher-initiated on Sarah's paper-and-pencil stories (TABLE 13). Sarah spent a long time on her story, "How the Earth Got Its Color", and although the story was short, she made a number of revisions, both independently initiated and teacher-initiated. The same was evident in her story, "Return of the Cubs".

TABLE 13

Independently Initiated and Teacher-Initiated

Revisions in Sarah's Paper-and-Pencil Stories

Initia	ted by:	
Student	Teacher	Total
21	18	39
32	22	54
53	40	93
57%	43%	
	21 32 53	21 18 32 22 53 40

On computer-composed stories, Sarah independently initiated 45% of the revisions and teacher-initiated revisions totaled 55% of the revisions (TABLE 14).

When Sarah began to write stories at the computer, she was observed to be enjoying the writing process much more and was anxious to get started each morning. She was also more enthusiastic about having conferences with her teacher. A change in attitude was evident when Sarah worked at the computer.

Independently Initiated and Teacher-Initiated
Revisions in Sarah's Computer Stories

	Initia			
Story	y Student Teacher		Total	
The Problem	25	39	64	
The Alley Cat	21 43		64	
Rusty	10	6	16	
The Clue	42	31		
Total revisions	98	119	217	
% of revisions	45%	554		

Joan's Revisions

independently initiated and 27% were teacher-initiated on her paper-and-pencil stories (TABLE 15). On her first story, "How Cats Eyes Glow", Joan chose not to conference with her teacher on her story. She had worked for a long time on it and had made many revisions on it. As she wrote her final draft for her

teacher she continued to make revisions as she recopied her story.

Independently Initiated and Teacher-Initiated
Revisions in Joan's Paper-and-Pencil Stories

Initia			
Student	Teacher	Total	
70	-	70	
39	41	80	
109	41	150	
734	27%		
	70 39 109	70 - 39 41 109 41	

Sixty percent of Joan's revisions at the computer were independently initiated and forty percent were teacher-initiated (TABLE 16).

Under both writing conditions Joan revised more independently than she did in conference with her teacher.

Independently Initiated and Teacher-Initiated Revisions
in Joan's Computer Stories

	Initia		
Story	Student Teacher		Total
Teenager's Knees	28	14	42
Goodbye, Sweet Val	ley		
High	48	37	85
Total revisions	76	51	127
t of revisions	60%	40%	
•	i .		

Summary of Findings of the Number of Independently Initiated and Teacher-Initiated Revisions

TABLE 17, summarizes the number of independently initiated and teacher-initiated revisions by the four students under both writing conditions.

Three of the four students made more independently initiated revisions than teacher-initiated revisions on their paper-and-pencil stories. The fourth student made more revisions that were teacher-initiated.

Percentage of Revisions in Both Composing Modes When
Independently Initiated and Teacher-Initiated

		Paper-a	ind-Pencil	Comput	Computer	
Student	-	*1/1	*T/I	1/1	7/1	
Michelle		664	34%	37%	631	
Heghan		46%	541	478	53%	
Sarah		57%	43 t	45%	551	
Joan		731	27%	60%	403	
λv	erage t	634	37%	47%	539	

- * I/I Independently Initiated
- * T/I Teacher-Initiated

When comparing independently initiated and teacher-initiated revisions done on computer stories, only one student made more independently initiated revisions. The remaining three students made more teacher-initiated revisions on their computer stories. Meghan made less independently initiated revisions under both conditions, while Joan made more independent revisions under both conditions.

When overall average percentages are compared, in the paper-and-pencil mode, there were more independently initiated revisions. In the computer mode, a greater percentage of the revisions were teacher-initiated. The paper-and-pencil revisions done independently by the four students were 63% of the total, and those initiated by the teacher accounted for 37% of the revisions. With the computer composed stories, a greater percentage of the revisions, 53%, were teacher-initiated and 47% were independently initiated by the students.

When the total number of revisions (independently initiated + teacher-initiated) were analyzed, three of the four students made more revisions on their stories composed at the computer than they did on the stories composed with paper and pencil. The fourth student made slightly more revisions on paper-and-pencil stories than on the stories she composed at the computer. The students, on the average, made less revisions on their paper-and-pencil stories (39%) than on their stories composed at the computer (61%) (TABLE 18).

Percentage of Revisions on Paper-and-Pencil Stories and
Computer Composed Stories

	Mode of Co	mposing
Student	Paper-and-Pencil	Compute
Michelle	439	574
Meghan	319	694
Sarah	30%	70%
Joan	54%	46\$
	Average t 39t	61%

Molistic Secring of Stories

The third research question asked if students produced stories of the same quality when using paper and pencil as compared to using a word processing program on the computer. To address this research question, a holistic scoring scale (Appendix 4) was used to determine the quality of the stories.

Two raters, using this holistic scoring scale, independently scored each of the stories written by the students. Reproductions of the students' stories, in word processing form, were presented to the raters. The stories were in random order; the raters did not know which stories were composed with paper and pencil and which were composed at the computer.

The scores, as assigned by the raters are found in TABLE 19 and TABLE 20.

On paper-and-pencil stories Rater 1 gave the students an average score of 2.44 on their stories. Rater 2 gave an average score of 2.22 on the same stories.

Rater 1 gave an average score of 2.73 out of 4 on the students' stories composed at the computer. Rater 2 gave the same stories an average score of 2.33.

In both paper-and-pencil stories and those composed at the computer, Rater 1 gave higher average scores than Rater 2, and both raters gave higher average scores to the stories composed on the computer than those written with paper and pencil.

When the scores assigned by both raters for each story are totalled and averaged, the average score assigned by the raters for stories composed at the computer was 5.06, compared with an average score of 4.67 for the stories composed with pencil and paper.

TABLE 19

Bolistic Scoring of Paper-and-Pencil Stories

		Rater		
Student Story	Story	1	2	1+2
Michelle	Candy Man	2	2	4
	The Fountain	2	2	4
	Time of Dinosaurs	3	3	6
Meghan	Babysitting	2	2	4
_	Earth Worm City	3	3	6
Sarah	Return of the Cubs	3	2	5
	How the Earth Got			
	Its Color	2	2	4
Joan	How Cats Eyes Glow	2	2	4
	Untitled Story	3	2	5
	Average	2.44	2.22	4.67

TABLE 20

Enlistic Secring of Computer Stories

		Rater		
Student	Story	1	2	1+2
Michelle	Mary Finds A Friend	3	3	6
	Sam	2	2	4
	Sick	3	2	5
	The Big Adventure	3	3	6
	Redhead	2	2	4
	Allen	2	2	4
leghan	Rosebud	3	3	6
	Toby the Turtle	3	2	5
	Mouse Eyes (Part 2)	3	2	5
Barah	The Problem	3	2	5
	Alley Cat	3	3	6
	Rusty	2	2	4
	The Clue	3	2	5
Toan	Teenagers Knees	3	3	6
	Goodbye Sweet			
	Valley High	3	2	5
	Average	2.73	2.33	5.06

Holistic Scores of Michelle's Stories

Michelle composed 3 stories using paper and pencil and 6 stories using the computer. The two raters gave Michelle identical scores on her paper-and-pencil stories. On the stories composed at the computer, 5 of 6 scores were identical, while 1 score differed by 1 point on the 4 point scale. Michelle's average score on the paper-and-pencil stories was 4.67 (TABLE 21) while her average score on the computer stories was 4.83 (TABLE 22).

Molistic Scoring of Michelle's Paper-and-Pencil Stories

		Rater	
Story	1	2	1+2
Candy Man	2	2	4
The Fountain	2	2	4
Time of Dinosaurs	3	3	6
Avera	ge 2.33	2.33	4.67

TABLE 22

Molistic Scoring of Michelle's Computer Stories

		r	
Story	1	2	1+2
Mary Finds a Friend	3	3	6
Sam	2	2	4
Sick	3	2	5
The Big Adventure	3	3	6
Redhead	2	2	4
Allen	2	2	4
Average	2.50	2.33	4.83

Holistic Scores of Meghan's Stories

Meghan composed two stories using paper and pencil and three stories at the computer. The two stories Meghan wrote with paper and pencil were given identical scores by the two raters (TABLE 23). On her computer-composed stories the raters gave one identical score, with the other two scores differing by only one

point on the scoring scale (TABLE 24). Meghan's average score on the paper-and-pencil stories was 5.00, while her average score on the computer stories was 5.33.

TABLE 23

Bolistic Scoring of Meghan's Pencil and Paper Stories

			Rater	
Story		1	2	1+2
Babysitting		2	2	4
Earth Worm City		3	3	6
	Average	2.50	2.50	5.00

Holistic Scores of Sarah's Stories

Sarah's stories consisted of two paper-and-pencil stories and four computer-generated stories. On the two stories composed with paper and pencil, the raters scores agreed on one story and differed by one point on

TABLE 24

Holistic Scoring of Meghan's Computer Stories

		Rater	
Story	1	2	1+2
Rosebud	3	3	6
Toby the Turtle	3	2	5
Mouse Eyes (Part 2)	3	2	5
Average	3.00	2.33	5.33

the second story (TABLE 25). Of the four stories composed at the computer, two of the four stories received the same score from the raters, while the other two scores differed by one point (TABLE 26). Sarah's average score, as assigned by the raters, was higher on the computer stories (5.00) than on the paper-and-pencil stories (4.50)

TABLE 25

Bolistic Scoring of Sarah's Paper-and-Pencil Stories

	Rater		
Story	1	2	1+2
Return of the Cubs	3	2	5
How the Earth Got Its Color	2	2	4
Average	2.50	2.00	4.50

Holistic Scores of Joan's Stories

Joan composed the same number of stories (two) with paper and pencil and at the computer. For Joan's two paper-and-pencil stories and for her two stories composed at the computer, the raters gave one identical score and differed by one point on the four point holistic scale for the second story (TABLE 27 and TABLE 28). Joan's average score on the paper-and-pencil stories was 4.50, while on the computer stories it was 5.50.

TABLE 26

Holistic Sooring of Sarah's Computer Stories

		Rater		
Story	1	2	1+2	
The Big Problem	3	2	5	
The Alley Cat	3	3	6	
Rusty	2	2	4	
The Clue	3	2	5	
Avera	ge 2.75	2.25	5.00	

Summary of Findings of Molistically Scored Stories

A comparison of the students' total average scores of paper-and-pencil stories and computer stories shows that all four students scored higher on the computer stories than those produced with paper and pencil (TABLE 29).

TABLE 27

Molistic Scoring of Joan's Paper-and-Pencil Stories

	Rater		
Story	1	2	1+2
How Cats Eyes Glow	2	2 2	4
Untitled Story Average	3 2.50	2.00	4.50

Michelle was identified by her teacher as a good writer. When her stories were holistically scored by the two raters, her scores were not higher than the other writers. In fact, she had the lowest average score on her computer-generated stories. Heghan, identified as an average writer, had the highest average score on her paper-and-pencil stories and on the stories composed at the computer. Sarah had been identified by her teacher as a weak writer. Her average holistic scores were not greatly different from the writers identified as good and average.

TABLE 28

Bolistic Scoring of Joan's Computer Stories

		Rater	
Story	1	2	1+2
Teenagers Knees Goodbye Sweet Valley High	3	3 2	6
Average	3.00	2.50	5.50

Quantity of Writing

The fourth research question asked if there was any difference in the amount of writing produced when writing with paper and pencil compared with the amount produced when using a word processing program. In order to address this research question, a total word count (TABLE 30) was done of the final product paper-and-pencil stories and computer stories written by each student.

Quality of Paper and Pencil and Computer Stories

	Average Holis	Average Holistic Scores		
Student	Paper-and- Pencil Stories	Computer Stories		
Michelle	4.67	4.83		
Meghan	5.00	5.53		
Sarah	4.50	5.00		
Joan	4.50	5.50		

When a word count was done, three of the four students produced more writing when using the word processor on the computer. Only Joan, wrote more when using paper and pencil to compose her stories.

Michelle had been identified by her teacher as a good writer. She produced the most text. She indicated in her interview that she liked to write, which was confirmed in the study. She produced almost twice as much text when using the word processor on the computer.

TABLE 30

Length of Paper-and-Fencil and Computer Stories

	Number of Words		
Student	Paper-and- Pencil Stories	Computer Stories	
Michelle	1 351	2 569	
Heghan	1 115	1 817	
Sarah	386	1 191	
Joan	884	651	
Total	3 736	6 228	

Meghan, described by her teacher as an average writer, also wrote more at the computer than with paper and pencil.

Sarah, described as a weak writer, who found writing to be a challenge, produced a small amount (386 words) with paper and pencil. A far greater amount (1191 words) was written at the computer by Sarah.

Identified by her teacher as an average writer,

Joan told me she only liked to write "sometimes". Joan

produced the smallest amount of text overall. At the

computer, she produced less (651 words) than she did with paper and pencil (884 words).

CHAPTER 5

CONCLUSIONS AND IMPLICATIONS OF THE STUDY

Chapter Overview

This chapter examines the findings of the study in relation to conclusions supported by these findings.

Implications of the findings for teachers and for future research in this field are also discussed.

Findings and Conclusions

This study investigated grade six writers as they composed and revised when using two modes - paper and pencil and a word processor on the computer. The types of revisions and quality and quantity of writing were examined. The findings and conclusions of this study are reported as answers to the research questions that guided the study.

Research Question 1

What similarities and differences exist in the types of revisions done by grade six students when writing with paper and pencil as compared with writing using a word processor on the computer?

Similarities were exhibited in the revision strategies employed by the student writers under both

writing conditions. The students made limited revisions in CONTENT, DEVELOPMENT, and STYLE categories when writing with paper and pencil and on the word processor at the computer. Whether the revisions were independently initiated or teacher-initiated did not affect the quantity of revisions in these categories - it was minimal.

During the interviews with the students, it was evident that they saw revisions as those changes that came first, under the CONVENTIONS category and second, under the SENTENCE SENSE category. Under both writing conditions, the students made the major portion of their revisions in these categories. The patterns associated with these two categories varied with each of the four student writers.

CONVENTIONS and STYLE were compared for both paper-and-pencil compositions and computer compositions. Three students made more revisions in SENTENCE SENSE than in the CONVENTIONS category on their paper-and-pencil stories. Three students made more revisions in the CONVENTIONS category than in the SENTENCE SENSE category on their computer stories. TABLE 31 shows the patterns exhibited by the student writers.

From the findings it may be concluded that students concerned themselves with the COMVENTIONS

category when revising stories written at the computer, and to a lesser degree, with the SENTENCE SENSE category.

Quantity of Revisions in SENTENCE SENSE and CONVENTIONS
Categories

Student	Paper-and-Pencil Stories		Computer Stories	
	*1	*2	1	2
Michelle	*88	*CON	5 \$	CON
Meghan	CON	88	CON	SS
Sarah	88	CON	CON	SS
Joan	88	CON	CON	SS

- *1. Largest number of revisions this category
- *2. Second largest number of revisions this
- category *SS SENTENCE SENSE category
- *CON- CONVENTIONS category

It may be speculated that students at this grade level are aware of the need for having their stories make sense and that the choice of words has begun to be important to them, as evidenced by the quantity of revisions in the SENTENCE SENSE category on paper-and-pencil stories.

One might also speculate that the process of revision on the computer has differences from the process of revision when writing with paper and pencil and that students may need more instruction on how to revise on the computer and how it differs from revision when composing with paper and pencil.

Research Question 2

How does the conference, an integral component of process writing, affect the quantity of revisions done by developing writers?

Few would argue the importance of conferencing in process writing. The findings of this study indicate that the role of the conference can be even more beneficial with stories composed at the computer. Stories composed at the computer had more teacherinitiated revisions than independently initiated revisions. The students, as a group, averaged more independently initiated revisions than teacherinitiated revisions on stories written with paper and pencil. The claim that clean copy may be deceiving to Student writers, perhaps, student writers has merit. do not as readily see the revisions that are needed because the story looks neat from the very first draft. Also to be considered is the teacher's own comments in the interview. She stated that perhaps she initiated

more revisions on stories composed at the computer because it was quicker and easier for the students. With paper-and-pencil stories, she was aware that they had to be rewritten and changes were not as easily made.

Individually, two of the writers exhibited the general pattern noted in the preceding paragraphs. One student made more teacher-initiated revisions than independently initiated revisions under both writing conditions. The fourth student made more independently initiated revisions than teacher-initiated revisions under both writing conditions.

The students made more revisions on stories composed at the computer than on those stories written with paper and pencil. Almost two thirds of all revisions were made on stories the students composed at the computer.

That the teacher, in conference, is an important part of the writing process has been acknowledged. From the findings it may be concluded that the combination of conferencing and the computer as the mode for composing are advantageous to young writers as they revise their compositions.

Research Question 3

Do students produce compositions of the same quality when composing with paper and pencil and when using a word processing program on the computer?

The compositions of the four student writers, when analyzed using a holistic scale, showed that the quality of writing was marginally better on stories composed at the computer, compared with stories composed with paper and pencil.

All four students received a higher average score on their computer compositions than on their paper-and-pencil compositions. This finding was consistent for both raters using an identical four point scale to score the students' compositions.

One factor which confuses the issue is the different input made by the teacher when the computer was used. The teacher stated that she initiated more revisions to the computer-generated stories because she felt the change could be made more easily. While others have reported that computer-generated stories are better, we are left with a question. Were the computer-generated stories better because of what the children did or were they better because of the teacher input? Clearly, this is a question that will need to be researched in greater depth.

These findings support the conclusion that the compositions produced on the computer by these children were of higher quality than those produced with paper and pencil. What is not known is how the phenomena occurred. It certainly appeared that the children enjoyed composing at the computer and were more likely to make changes to their stories.

Research Question 4

Is there any difference in the quantity of writing produced when writing with paper and pencil when it is compared with the amount produced when using a word processing program on the computer?

Three of the four students produced a greater quantity of writing when composing at the computer than when using paper and pencil to write their stories.

The difference in quantity produced in the two modes during the composing process varied with each student. The student identified by her teacher as the weakest writer produced almost three times as much writing on the computer as she did when she wrote with paper and pencil. When using the computer to compose, the good writer produced almost twice as much as when using paper and pencil to compose.

Of the two average writers, one produced one third more writing using the word processor and the other

produced one quarter more when writing with paper and pencil.

While the present study does not provide a definitive answer to why the children composed more at the computer, each child in the study indicated that she preferred writing at the computer and each reported that she enjoyed writing more when using the computer. There are grounds for speculating that the computer may motivate many children to write more.

Implications for Teaching

While the conclusions of this study require more research with larger numbers and different kinds of children, the findings are consistent with a number of other studies in this area of research. Many children do seem to both prefer and enjoy composing at the computer. If more writing leads to better writing, then teachers can feel more confident in introducing the children to composing at the computer, even if time needs to be taken first to have the children acquire keyboarding skills.

The quality of the students compositions was found to be better on those composed at the computer. This, coupled with the positive attitude students have about using computers, has implications for the teaching of writing. All four students in this study indicated

that they would compose only at the computer, if given the opportunity. Over time, would the enthusiasm wane as the novelty of using the computer wore off? Would the consistent use of the computer in process writing contribute to improved quality of writing? These questions remain to be answered.

The word processor appears to encourage increased revision when the conference is used in the writing process. However, under both writing conditions, the revisions were similar. The types of revisions done by the students showed no major differences when paper-and-pencil compositions were compared to computer compositions. Strategies of revision do not change or necessarily become better by using the computer. These strategies can only be acquired by modelling and instruction by the teacher and practice by the student.

The students in this study made more revisions independently with paper and pencil than they did when writing on the computer. Though students like the clean copy that they have from the first draft of their writing, perhaps this clean copy makes them less able to see the need for revision, whereas the handwritten work appears to be less polished and more in need of revisions. New skills need to be taught in how to revise when composing at the computer.

Implications for Future Research

Further study would be necessary to confirm the findings and generalizations gleaned from this study. From the small number of subjects and the relatively short period of time for the study I would not presume to declare these findings as unequivocal.

Future inquiry might investigate questions that arose, but were not within the boundaries of this study. A similar study may be undertaken with students at the same grade level or other grade levels.

In this study, an attempt was made to have students record on tape the reasons they made changes in their stories. The students found this difficult and while they were encouraged to do this, it was not done consistently by all four subjects. The 'think aloud' protocols may be a valuable tool in learning during the writing process. Students who have been instructed in 'think aloud' protocols and use them as part of their daily writing may contribute valuable information to the growing body of research available about process writing.

This study took place over a relatively short time period. A longitudinal study would undoubtedly provide more conclusive data about how students write and revise under both writing conditions.

The students in this study had previous, but not extensive experience, with keyboarding and with the word processing program. They had never composed directly at the computer before the study. A study of students who had composed at the computer for a long period of time would reveal more answers to the questions that guided this study.

The student in this study that was identified by her teacher as a weak writer produced three times as much writing when using the word processor for composing. A study involving students with difficulties in the writing process may provide directives for instruction to improve the skills of these writers.

This study did not address the question of gender differences. A study including male subjects may show differences from the findings of this study, which analysed only the writing of female grade six students.

Other topics that might be considered for study are the relationship between quality and quantity in process writing, the impact which prompting programs may have when combined with conferencing in process writing, and the role of the teacher in process writing using the computer as the writing mode.

Before closing, another issue that needs to be discussed is the numerical aspect of the rating system

used in this study and in many others. Clearly, the changes identified in this study as CONVENTIONS and SENTENCE SENSE involve lower order changes, which are more likely to occur with greater frequency. CONTENT, DEVELOPMENT, and STYLE involve idea level changes and are more likely to occur with less frequency. When a rating system depends simply on counting 'changes' as per each category it would be rare indeed for the lower order changes not to outnumber the higher order changes. If we are to have a more precise understanding of writing revisions it may be more fruitful to assign different numerical ratings to the categories which would help balance the frequency factor in comparing the types of revisions made by writers.

Concluding Statement

This exploratory study investigated grade six students as they were involved in process writing in the natural, instructional setting of their classroom. Writing on topics of their own choice, the four students composed in two modes: paper and pencil, and using a word processing program on the computer. Revisions, both independently initiated and teacher-initiated, were analyzed and the final compositions were examined for quality and quantity.

The 1990's promise to be even more exciting than the eighties as we are exposed to the newest computer technology. The challenge for teachers as we approach the twenty-first century is to capitalize on the interest and enthusiasm of our students for this technology by fully integrating it into our students' learning experiences.

BIBLIOGRAPHY

- BALAJITHY, E. MCKEVENY, R., LACITIGNOLA, L. Microcomputers and the Improvement of Revision Skills. The Computing Teacher: Dec-Jan: 28-31, 1986-87
- BEAN, J.C. Computerized word processing as an aid to revision. College Composition and Communication: 34(2), 146-148, 1983
- BEACH, R. BRIDWELL, L. New Directions in Composition Research. New York, NY: The Guilford Press, 1984
- BRANAN, K. Moving the writing process along. Learning: 13(3), 22-26, 51, 1984
- BRIDWELL, L.S. Revising strategies of twelfth grade students' transactional writing. Research in the Teaching of English: 14, 197-222, 1980
- CALKINS, L. M. Notes and comments: children's rewriting strategies. Research in the Teaching of English: 14(4), 331-341, 1980
- CALKINS, L. M. Lessons from a Child: On the Teaching and Learning of Writing. Exeter, NH: Heinemann Educational Books, 1983
- COLLIER, R. M. The word processor and revision strategies. College Composition and Communication: 34(2), 149-155, 1983
- COLLINS, J. L. SCHEERS, E. A. Writing On-Line: Using Computers in the Teaching of Writing. Upper Montclair, NJ: Boyton/Cook Publishers, Inc, 1985
- COOPER, C. R. ODELL, L. (Eds.). Research on Composing: Points of Departure. Urbana, ILL: National Council of Teachers of English. 1978
- CRAMFORD, R. Writing, Word Processing, and Grade Four Writers. Unpublished master's thesis, University of Alberta, Edmonton, Alberta, 1989

- DAIUTE, C. Word processing: can it make even good writers better? Electronic Learning: 1(4), 229-31, 1982
- DAIUTE, C. Writing, Creativity and Change. Childhood Education: 59(4), 227-231, 1983
- DAIUTE, C. Writing and Computers. Reading, MASS:
 Addison-Wesley Publishing Company, 1985
- DAIUTE, C. Physical and cognitive factors in revising: insights from studies with computers. Research in the Teaching of English: 20(2), 141-159, 1986
- DE ALMEIDA-VAN HOOYDOOK, J. Revision Processes
 Utilizing Word Processors. Unpublished
 master's thesis, University of Alberta,
 Edmonton, Alberta, 1986
- EDMONTON PUBLIC SCHOOL BOARD. Holistic Scoring Guide. Mimeographed material from Writing Evaluation Workshop, October, 1986
- EMIG, J. The Composing Processes of Twelfth Graders. Urbana, ILL: National Council of Teachers of English, 1971
- FAIGLEY, L. WITTE, S. Analyzing revision. College Composition and Communication: 32, 400-414, 1981
- FAIGLEY, L., CHERRY, R., JOLLIFFE, D. SKINNER, A. Assessing Writers' Knowledge and Processes of Composing. Norwood, NJ: Ablex Publishing Company, 1985
- FISHER, G. Word processing-will it make kids love to write? Instructor: XCII(6), 87-88, 1985
- FLOWER, L. HAYES, J. R. The cognitive process theory of writing. College Composition and Communication: 32, 365-387, 1981
- FREEDMAN, S. (Ed.). The Acquisition of Written
 Language. Norwood, NJ: Ablex Publishing
 Corporation, 1985

- GENTRY, L. A. What Research Says About Revision.
 Washington, DC: National Institute of
 Education. (ERIC Document Reproduction
 Service No. 225 413), 1982
- GRAVES, D. H. Research update: what children show us about revision. Language Arts: 56 (3), 312-318, 1979
- GRAVES, D. H. Writing: Teachers and Children at Work. Exeter, NH: Heinemann Educational Books, 1983
- GRAVES, D. H. HANSEN, J. The author's chair. Language Arts: 60 (2), 176-183, 1983
- GREEN, J. O. Computers, kids and writing: an interview with Donald Graves. Classroom Computer Learning: 4 (8), 21-22, 28, 1984
- HARRIS, J. Student writers and word processing: a preliminary evaluation. College Composition and Communication: 36 (3), 323-330, 1985
- HARSTE, J. C. WOODWARD, V. A. and BURKE, C. L.
 Language Stories and Literacy Lessons.
 Exeter, NH: Heinemann Educational Books,
 1984
- HAWISHER, G. The Effects of Word Processing on the Revision Strategies of College Students. Paper presented at the annual meeting of the American Educational Research Association, San Francisco, CALIF., APRIL, (ERIC Document Reproduction Service No. ED 268 548), 1986a
- HAWISHER, G. Studies in Word Processing. Computers and Composition: 4 (1),6-31, 1986b
- HOPKINS, S. CLARK, S. Videotaping Software. The Computing Teacher, June, (3), 56, 1986

- KANE, J. H. Computers for Composing. Paper presented at the annual meeting of the American Educational Research Association, Montreal, QUE. (ERIC Document Reproduction Service No. ED 230 978), April, 1983
- KOPP, R. J. The Effects of a Word Processor on the Revision Strategies of Grade Five Writers When Composing in the Narrative Mode. Unpublished master's thesis, University of Alberta, Edmonton, Alberta, 1985
- KRASHEN, S. D. Writing: Research, Theory and Application. Elmsford, NY: Pergamon Press Inc., 1984
- LARTER, S. Writing with Computers in the Elementary Grades. Toronto, ONT: MGS Publications Services, 1987
- MONAHAN, B. Revision strategies of basic and competent writers as they write for different audiences. Research in the Teaching of English: 18(3), 288-304, 1984
- MOROCCO, C. You Changed My Story! The Word Processor as an Intelligent Environment for Revision. Paper presented at the annual meeting of the American Educational Research Association, San Francisco, CA, 1986
- MURRAY, D. In COOPER, C.R., ODELL, L. (Eds.)
 Research on Composing. (pp. 85-103).
 Urbana, ILL: NCTE, 1978
- MYERS, M. A Procedure for Writing Assessment and Holistic Scoring. Urbana, ILL: NCTE, 1980
- NATIONAL

 ASSESSMENT OF EDUCATIONAL PROGRESS
 Write/Rewrite: An Assessment of Revision
 Skills: Selected Results from the Second
 National Assessment of writing
 Washington, DC: U.S. Government Printing
 Office. (ERIC Document Reproduction
 Service No. ED 141 \$26), 1977

- NOLD, E. In GENTRY, L. A. What Research Says
 About Revision. Washington, DC: National
 Institute of Education. (ERIC Document
 Reproduction Service No. 225 413), 1982
- PAPERT, S. Mindstorms: Children, Computers and Powerful Ideas. New York: Basic Books, 1981
- PEACOCK, C. Teaching Writing. New Hampshire: Croom Helm Ltd, 1986
- PERL, S. Composing processes of unskilled writers. Research in the Teaching of English: 13 (4), 317-336, 1979
- PHENIX, J. HANNAN, E. Word processing in the grade one classroom. Language Arts: 61 (8), 804-812, 1984
- RODRIGUES, D. Computers and Basic Writers. College Composition and Communication: 36 (3), 336-339, 1985
- ROSS, J. In COLLINS, J. L. and SCHMERS, E. A. (Eds.) Writing On-Line: Using Computers in the Teaching of Writing. Upper Montclair, NJ: Boyton/Cook Publishers, Inc, 1985
- SCHANK, E.T. Word Processor versus 'The Pencil' Effects on Writing. Unpublished master's thesis, Kean College of New Jersey. (ERIC Document Reproduction Service No. ED 270 791), 1986
- SCHWARTZ, H. Teaching writing with computer aids. College English: 46 (3), 239-247, 1984
- SCHWARTZ, M. Computers and the teaching of writing. Educational Technology: 22 (11), 27-29, 1982
- SEKULER, R. From quill to computer: the computer has done away with writer's cramp, but can it do anything about writer's block?
 Psychology Today: February, 37-39, 42, 1985

- SMITH, N.J. The word processing approach to language experience. The Reading Teacher: February, 556-559, 1985
- SOLOMON, G. Teaching Writing with Computers. Englewood Cliffs, NJ: Prentice-Hall, 1986
- SCHMERS, N. Revision Strategies of Student Writers and Experienced Writers. College Composition and Communication: 31 (4), 378-388, 1980
- STANGER, D. Magic Slate. Pleasantville, NJ: Sunburst Communications, 1986.
- WHEELER, F. Can word processing help the writing process? Learning, 13 (7), 54-60, 1985

APPENDICES

APPENDIX 1 LETTER TO PARENTS

#9-10453-20 Ave. Edmonton, Alberta T6J 5Al April 9, 1987

Dear Parent,

I am a teacher who is now studying at the University of Alberta. From April 13 to approximately June 5, 1987 I will be at ******* School to do a research study. The study will be about computers and the kinds of stories children write when they use the computer compared with the kinds of stories they write when they use pencil and paper. I would like your permission for your child to participate in my study.

I will be using video equipment to record the children's writing at the computer and at their desks. The video will not focus on their faces. It will focus on the actual writing because I need samples of the children as they actually write and make changes to what they write.

This research project has been approved by the University of Alberta, ******** County, Mrs. **** and Mrs. ******. I need your permission for your child to take part in this study. If, for any reason it becomes necessary, your child may withdraw from the study at any time.

Thank you for taking time to complete the form below. I appreciate your cooperation. Should you have any questions, please feel free to contact me at 438-1816.

Yours sincerely,

	Pamela Carter
I (do, do not) give my	permission for my child,
	, to participate in the study to be
done by Pamela Carter	at ****** School, commencing April 13,
1987.	
	Signature of Parent or Guardian

APPENDIX 2 STUDENT INTERVIEW DATA

Michelle

Attitude About Writing

Michelle was described by her teacher, Mrs. Taylor, as a pleasant, cheerful girl who was also "keen, interested and enthusiastic" and an above-average student who "has done well this year". She indicated that Michelle enjoyed writing, but was reluctant to revise. In part:

Michelle has always had very good writing ability. Her attitude was always good... She always has good story ideas. She has never been one to sit and say, 'I can't think of anything to write'. She doesn't care to revise. She (Michelle) would say 'this is good enough. Here is my idea, I want to go on with something else'.

In an interview with Michelle following the study, she confirmed the teacher's belief that she liked to write. She explained that she liked to write "because you get all those interesting ideas in your head and you just can't tell them to someone, so you have to write them down on paper". Michelle liked to write about "stuff that pops into your head . . . make-believe stuff or anything". She didn't like to write about gore. She felt that the easiest part of writing was knowing what she was writing about and "finding the plot of the story".

The hardest part of writing, Michelle told me, was when she got stuck on something and she couldn't think of a word to go next. She added that it was sometimes hard to get started on a story.

Usually it is the beginning, I have all these ideas and I can't choose which one. Then I am in the middle of the story, and I have used all my ideas and I can't find anything. Then at the end I need one more sentence to complete it, to make it perfect, and I can't find it.

Michelle added that sometimes she got stuck at the beginning, middle and end, sometimes on just one or two of these parts and sometimes not at all.

Getting Ideas and Choosing Topics for Stories

Although Michelle said she did not read much, she got a lot of her ideas for her stories from the books she did read. Other sources of topics are TV, a movie, a picture on the wall, or from something someone has said. Michelle commented that she had trouble choosing a topic when she first was composing at the computer, due to the change in format from the pencil and paper mode but that she had now "gotten used to the computer".

Michelle preferred to choose her own writing topics, rather than have them assigned by the teacher. The reason she felt this way was because she found that it often sounded like a good topic, but "you get the first few sentences done and no matter what you do you can't think of anything else". Michelle's teacher usually let the students choose their own topics for writing, but had a series of story starters on cards that students could use when they were stuck for an idea. "Sometimes I will use those cards that give you ideas, or I will think and think," Nichelle told me.

Sharing the Stories

Michelle was not as enthusiastic about sharing her writing. She didn't like to share in front of the class and preferred to share with one or two people. In Mawthorne School, on Friday afternoons, the whole school came together to share published stories and Michelle did not mind sharing in front of the whole school. When I questioned Michelle for a reason about her preference to share in front of the whole school, she replied that the whole school was bigger. In class sharing, students made comments and asked questions, but during school sharing on Friday, "they just clap," said Michelle.

Making Mevisions in the Stories

When I asked Michelle about changes (revisions) she made, she immediately mentioned editing changes such as commas, followed by changes so that sentences "made sense". She reported it was easier to make changes on the computer and liked not having to "scribble out" (though she also claimed it did not bother her to scribble out) and felt she would make more changes using the computer. "You can just flip

back to any part of the story and take out a couple of words and add a couple there", something she didn't do when using a pencil when composing her stories. She said she liked to leave stories and go back later to "correct and conference", because then the stories seemed new to her.

Michelle pointed out that she used the screen for most of the revisions she made. She did not feel that having only a part of the story on the screen was a problem for her. She indicated that she did read the printed copy of her story, but more frequently used the screen for her revisions.

Student Attitude About Composing Modes

Prior to the study, Michelle had typed in stories that she had hand written, but had never done any composing at the computer. After her experiences composing at the computer, Michelle did not hesitate to choose the computer as her preference for her story writing. She found it "fun" and "interesting" to write at the computer.

Use of the Word Processing Program

I observed that Michelle typed quickly, though this caused her to sometimes omit a letter. Michelle was able to use the correct keyboard ingering, but usually reverted to the two-finger F.:hod. She was able to use the Magic Slate word program with a minimum of difficulty, being the to make a new file, get a previous file on the screen, use a variety of the editing capabilities of the program, and save the work she had done.

Meghan

Attitude about writing

Meghan's teacher, Mrs. Taylor, reported that Meghan was in the average range of achievement for her grade. She further commented that Meghan had a positive attitude about school, was willing to try anything, and

got along well with her classmates.

Mrs. Taylor found Meghan's writing ability to also be within the average range, but noted that she had problems with spelling and grammar. She had lots of ideas, some of which her teacher observed were "very simplistic". The teacher also believed that Meghan was willing to work at revising her stories. Although Meghan normally wrote a lot, Mrs. Taylor felt Meghan had written more during the study, especially when composing at the computer.

For the purpose of the study, the students were encouraged to conference each story with the teacher. Meghan told both the teacher and me that she liked the idea of the more frequent student-teacher conferences.

Meghan liked to write "because it is fun and you can get out everything you think about". She liked to write about people, things that might happen, as well as make-believe. Starting a story was the easiest part of writing for Meghan. She thought that solving a problem was the hardest part of writing, and found that sometimes it was hard to write an ending for a story.

Getting ideas and choosing topics for stories

Ideas for Meghan's stories came from "stuff that happens to me, sometimes it is from dreams, or sometimes it is from my sister's stories. She writes good stories". When she had difficulty thinking of an idea to start a story, she would sometimes "block things out and let my imagination run away. Sometimes I talk to my sister".

Meghan liked to read and reported that the books she read gave her ideas for her story writing. This was confirmed in her writing during the study. Mer idea for the story about babysitting grew out of having read a series of books about babysitting and having her first babysitting job the previous weekend.

Choosing her own topics to write about was Meghan's preference, but she added that there were times when it was easier if the teacher chose the topic. "When we were doing picture stories, it was pretty easy because she (Mrs.Taylor) gave us ideas and stuff".

Sharing the stories

Sharing was an enjoyable part of the writing process for Meghan and in response to a question about whom she liked to share with she replied:

If I think I wrote a good story, I would like to share Fridays (when the students share with the whole school). If I write an OK story, then with the class. If I am starting out, with a small group, but it ends up a good story then I would share.

Making revisions in the stories

Checking to see if the sentences made sense, spelling and paragraphing were Meghan's perception of revision. By paragraphing she meant indenting each time there was a new speaker. Meghan believed it was important to make changes in her work and explained it in this way:

If you write something and then you say this is done, and then if you read it over later you find spelling mistakes and punctuation mistakes.

Student attitude about composing modes

The difference Meghan noticed when composing at the computer and composing with pencil and paper was the number of typing errors she made when she was using the computer. She found it easier to make changes on the computer and liked the neat, tidy copy the printer produced for her. Meghan did not see a big advantage for one mode of composing over the other, but given the choice, would compose at the computer because she liked computers and thought they were "neat".

Use of the word processing program

Meghan was able to use the basic functions of the word processing program - open a new file, call up and

save a file, and use format commands such as centering and vertical spacing. She became more confident in the use of the program as the study progressed.

Although aware of the correct fingering for the keyboard, like her classmate Michelle, Meghan often

Although aware of the correct fingering for the keyboard, like her classmate Michelle, Meghan often usually used the two-finger method. She commented during the interview with me that she made more typing errors (spacing and punctuation) when she worked on the computer, compared to the number of similar errors when writing with her pencil.

Sarah

Attitude about writing

Sarah was determined, by Mrs. Taylor, to be a "low" student. Although Sarah read a lot, her teacher explained that it didn't "seem to transfer into her spelling skills, her speech, language skills, or writing. A lot of the time she is stuck with a topic. She doesn't seem to have much to say, even in her diary". Mrs. Taylor believed Sarah liked to write and pointed out that she was very creative in poetry writing, which was not included in this study. Sarah was "not afraid to work" in her teacher's estimation; however Mrs.. Taylor felt that Sarah didn't see or understand the point of some of the revisions.

I asked Sarah if she liked to write, to which she replied simply, "sometimes". Throughout the study there were several comments on tape that she made while working when she stated that she did not like to write.

She told me that the topics she liked to write about were horses and animals. For Sarah the easiest part of writing was "working on the computer". She found getting started hard in writing, as well as when it "didn't make sense and it is all mixed up".

Getting ideas and choosing topics for stories

Sarah got most of her ideas for stories from "TV shows, books, and things at school". During the study I frequently noticed Sarah having difficulty getting started on a story. In the interview following the study, Sarah confirmed that this was indeed a challenge for her.

I asked Sarah if she liked to choose her own topics for writing. Her initial reply was "I don't know". A little later in the interview she told me that most of the time she would prefer her teacher to assign the writing topic and only liked to choose her own topic "when something happens that I want to tell".

Sharing the stories

Sarah preferred to share with only one person. She did not care to share either in class or with the whole school on Friday afternoon. She felt shy about sharing with a group of any kind. She stated in the interview, "I don't like everybody staring at me" and

that when she shared with a group she always felt "scared".

Making revisions in the stories

Sarah told me it was important to make changes if a story didn't make sense. She felt "OK" about making changes depending on how she felt. Changes in spelling and sentences were the reasons Sarah mentioned.

Making changes on the computer was easier Sarah found because "it is easier to go back and not scribble out errors". She found that when she scribbled out she would get all confused and it looked messy. In her pencil and paper stories, Sarah copied one story over three times because she did not feel happy about how it looked.

Student attitude about the composing modes

Following the study, Sarah stated a preference for writing on the computer because:

- (1) it is more interesting
- (2) it doesn't take so long on the computer
- (3) you don't have to write it over again
- (4) it is faster.

Sarah could not think of any advantage for using the pencil, rather than the computer to write her stories. Given the opportunity, Sarah would use the computer all the time.

Use of the word processing program

Sarah's skill in using the word processing program improved as the study progressed. It did cause some difficulties for her, especially in formatting techniques (centering, vertical spacing). She was able to make a new file, call it up and save it without any difficulty. Her keyboarding skills were not as well developed as some of the other students in the study.

Joan

Attitude about writing

Joan was an "average student", said her teacher, who was always cheerful and cooperative in class. In her teacher's opinion, Joan had a good attitude about writing. Her comments about Joan included:

She has some grammar problems when she talks and when she writes. . . She has a lot of ideas with her writing, but again they are very simple ideas. She normally produces (when she writes).

During the interview, Joan's answer to the question about whether she liked writing was: "sometimes I do and sometimes I don't. It depends on what kind of mood I'm in." Joan liked to write about things that she knew. For her, the difficult part of story writing was finishing it.

Getting ideas and choosing topics for stories

Ideas for stories sometimes came from Joan's thoughts. Experiences often provided her with ideas. "Last year I took a skating test and I wrote about that." When Joan had difficulty thinking of an idea for a story, she would write a list of topics and then choose one for a story; however she found that she didn't need to do this very often.

Joan preferred to choose her own topics for writing. When she was "stuck for an idea" she liked to have her teacher's help.

Sharing the stories

Joan enjoyed sharing stories with her friends. She liked to share with the class "sometimes, but not all the time". She enjoyed the Friday afternoon sharing with the whole school. Just as Michelle had indicated, Joan liked Friday afternoon sharing because everyone just listened and didn't ask any questions.

Making revisions in the stories

Before conferencing with her teacher, Joan sometimes made changes in her stories. In response to

my question about what changes she made, she answered: "I look for capitals, commas, periods and spaces . .

.and spelling errors".

Making changes was important, thought Joan,
"because if you go and do your rough copy and don't look it over, when you do your good copy everything is going to be messed up". She felt that making changes was fun and that most of the changes she made were in paragraphs and spelling.

Student attitude about composing modes

Joan liked the computer better for composing her stories and found it easier to make changes with the computer because "it is more fun on the computer and you learn stuff".

The main difference Joan noticed when composing with pencil and paper compared to composing on the computer was that when she used her pencil her hand got really tired. It was easier to write using the computer mode and Joan felt that if she used the computer all the time she would write more. The computer was her choice of writing tool.

Use of the word processing program

As with the other students in the study, Joan's skill in the manipulation of the word processing program increased throughout the study. She was able to use the general features of calling up, opening and closing files, as well as the formatting commands of spacing, centering and typestyles.

Joan usually used the correct keyboard fingering

when composing her stories.

Summary of Interview Findings

The students varied in their attitudes about writing. Michelle and Meghan were positive and liked to write. Joan and Sarah were not as enthusiastic about the writing program and indicated they liked to write "sometimes". There was evidence that writing was difficult for Sarah. A variety of areas were cited as the easiest part of writing for these four grade six students. There was more consensus on the hardest part of writing. It was either starting the story or solving the problem (finishing the story).

Ideas for stories came from books, television, movies and the students 'own experiences. Three of the four students preferred to choose their own topics when they wrote. The fourth, Sarah, preferred to have

teacher-chosen topics.

When sharing stories, the students preferred to share in small groups or with the whole school on Friday afternoon. Only Meghan indicated that she really liked to share her stories, with the other three students having reservations about sharing. Sharing in class was the least favored, because the other students then asked questions.

The revision process, as perceived by the students, was basically editing procedures and changes so that their stories made sense. The students in the study found this to be easier and quicker on the

computer.

The attitude of the students was most positive about the use of the computer. All four would use the computer for composing stories, if given the choice. The reasons that they would make this choice was that using the computer was quicker, easier, and less tiring. The computer was fun and interesting.

The students had been given lessons and practice in keyboarding. Two of the students used the correct fingering most of the time, but the other two students used more of a "hunt and peck" method. They had used the word processing program Magic Slate to enter stories written originally with paper and pencil, but had not actually composed at the computer. At the beginning of the section of the study where they used the computer, each of the students showed some difficulties with using some of the commands in the program. With daily use of the program, their skills improved quickly. Three of the four students became fairly competent in the use of the program. The fourth, Sarah, improved but still need help fairly frequently, especially with formatting commands.

APPENDIX 3 TEACHER INTERVIEW DATA

Mrs. Taylor

The writing process and revision

Mrs. Taylor had a very positive attitude about the computer's capabilities as a writing tool, and was a computer user herself. The students had produced a school newspaper each term, using Newsroom, a desk top publishing program for elementary and junior high school students. Along with articles about the school and its activities, student writing was also a part of

the publication.

As part of process writing, Mrs. Taylor explained that she had worked on developing revision strategies during the year. She used a conference sheet and basically looked at each sentence to see if it made sense. She called it "a very basic thing . . . it wasn't in any great detail". When conferencing with students she had each one consider whether he/she had included all the information that was needed and removed anything that wasn't necessary. She tried to have the students look at the story as a whole.

Mrs. Taylor believed that the students perceived revision as "a lot of work and they would rather not do it". For most students, their idea of revision was fixing spelling mistakes, added Mrs. Taylor. She believed the student-teacher conference to be a crucial part of the writing process, adding that the student usually did not see the problem until attention was drawn to it by the teacher.

In the interview, Mrs. Taylor commented that it was possible she had initiated more revisions on the computer composed stories than with the pencil and paper stories. Her reason was that when using the pencil to compose

I unconsciously thought, this has to be rewritten, so it is good enough the way it is. With the computer I would think, oh this can be easily changed. . . So that could have been a subconscious thing I did. I think there were more revisions on the computer.

Comparing the two modes of composing

Mrs. Taylor thought composing at the computer was much easier for the students, once they were

comfortable with the word processing program. With experience, she believed they would see the value of the computer for revisions and start to make more than editing changes. She further stated that she thought the students "would really develop into good writers if they had it (the computer) continuously and didn't have pencil and paper, but just the computer".

Not having to rewrite their stories and ease on making changes were the strengths Mrs. Taylor believed were attractive to the students. Once students had good keyboarding skills, Mrs. Taylor saw no advantage for using pencil and paper during the writing process.

APPENDIX 4 BOLISTIC SCALE

molistic scale

- (4) The paper reads smoothly. The content is clear, well developed and well organised. The intent of the writer is clear and he/she shows an awareness of the audience. There is evidence of precise vocabulary and usage appropriate to the writing task. Mechanical errors are few and do not interfere with the meaning intended by the author.
- (3) The paper reads smoothly but may contain a few awkward parts.

 The content is somewhat clear and shows some organizational problems.

 The writer shows some awareness of the reader.

 There is some evidence of precise vocabulary but sometimes it may seem stilted or inappropriate to the task.

 There may be some mechanical errors but these do not significantly interfere with the meaning.
- (2) The paper reads unevenly.
 The content is not clear, although there may be some evidence that the writer knows what he/she means to say.
 Vocabulary usage is correct but does not convey the author's meaning specifically.
 Mechanical errors do interfere to some degree with the message.
- (1) The paper is difficult to read but some understanding of the author's meaning can be deciphered. The writer does not have an awareness of communicating meaning to a reader. Mechanical errors and sentence problems interfere significantly with communication.
- (0) The paper is not understandable. The writer lacks the ability to communicate meaning or write sentences.

APPENDIX S STUDENTS' STORIES

Michelle

Story 1 - Paper and Pencil

Candy Man

One day there was a kid named Joey. He had a sweet tooth. He would eat sandy and shocolate bars, and every kind of sweet he sould think of. One day Joey went to the sandy store to buy some shocolate bars, but they were all out. So then he went to where all the other sandles were, but there weren't any there either. Joey went to the manager to complain about their sandy selection. The manager told him that he was ordering in some more sandy.

Joey came back to the store 1 day later, but there still wasn't any candy of any kind. Joey had a fit and went to speak to the manager again. The manager told him not to get too upset, but Joey was yelling anyways. Joey finally got onto the managers nerves, so the manager salled in a pollogman to take Joey out.

The next day Joey went to a new store, but could only find some gum. He asked somebody why they didn't have any shocelate bers. The person told him that they hadn't getten them in yet and that they should be in, in 2 days. Two days went by and Joey went back. They had all the condice that Joey wented. Joey brought ten dollars that day. He bought gum, shocelate bers and really sweet

candles. Joey almost bought all the candles in the store. The only candles that were left were two gum balls. Joey went home for 2 cents so he could by the gum. When he came back they were gone. Joey saw the kid that bought them.

The boy turned around and asked how much the big 4 foot candy cost. The person said that she didn't know because she hadn't ever seen it before. She called out the manager and asked him. He said that he didn't remember ordering that before. Then he said that it would probably cost about \$6.00. The little boy bought the candy, and ate it.

Although Joey had been eaten by a boy, he thought, "If I ever become a boy again, I swear that I will never eat another sweet as long as I live!!"

Michelle

Story 2 - Paper and Pencil

The Fountain

One day Cindy went to the park to play with her friends. She saw a fountain that she hadn't seen before. She walked around it and behind a bush she found a door. Cindy walked in.

"Hello!" she said, but nobody answered. She walked in farther. Cindy sew three hallways with three signs. One sign said, "Man Eating Dregon." The second one said, "Walk This Way." And the third one said, "Fir Breathing Chicken." Cindy went down the one that said, "Walk This Way."

Cindy found that the hallway was desip and dismal, but she kept on walking. Finally she came to another door. She knocked on it.

"Yes." said a voice.

"May I come in?" asked Cindy.

"Maybe." answered the voice. Cindy decided to Just open the door.

"Hello!" she said again. "Is anybody home?" asked Cindy.

"Depends who your looking for." said the voice. Cindy walked into the resm. She looked around to see if she

gould find anybody. Then she heard a POOF!!! She turned around and there stood a person dressed in Blue.

"Who are you ? asked Cindy.

"I'm the one and only Humphrey!" And who might you be?

"Oh! Hy name is Cindy."

"And what are you doing here? asked Humphrey again.

. I saw the door and decided to come in." Cindy said.

"Well, why didn't you so down the hallway that said.
"Fire Breathing Chicken?"

"Because," replied Cindy, "I wanted to come down this way." Soon Humphrey and Cindy became good friends.

Humphrey invited Cindy over for dinner that night, bit Cindy said that she was going out that night, so Humphrey invited her over for the next night. She said "Okay!"

The next day Clady went to the park to go and visit Humphrey. She walked down the hallway toward Humphrey's place. She knocked on the door.

"Humphrey!" yelled Cindy. "are you home?" Humphrey didn't answered the door. Cindy decided to just walk in. "I'm here Humphrey!" she yelled again.

"Don't come int" Cindy heard a voice may.

"Is there seemething wrong Humphrey?"

"Maybel" he andrered.

"Well, I'm coming in wheather you like it or not." She walked into the kitchen and sat down. "Come on out Humphrey, don't be embarassed!" Humphrey slowly came to the kitchen door.

"Humphrey!" yelled Cindy, "What happened to your hair?"
"Hy hair?" questioned Humphrey.

"Yes," replied Cindy "You're bald!"

"Baid!" yelled Humphrey.

"Yes!" The Cindy and Humphrey talked a little about Humphrey having no hair. Cindy decided to go and buy Humphrey a wig.

Michelle

Story 3 - Paper and Pencil

Time of Dinosaurs

"Since the beginning of time, dinomates have roamed the earth in search of food." "Can I take a break from reading now Mrs. Green?" asked Charlie. "Just one more sentence." said Mrs. Green. "Theyv'e always been big animals." "There, that's it! Can I go out and play now?" asked Charlie. Mrs. Green nodded her head.

Later on, on the ride home from school, Charlie thought, "I'd like to go back in time. In the time of dinosaurs!" It was Charlie's stop now. Charlie got off the bus. He kept thinking about what he thought of before.

When Charlie got home, he went up to his room and pulled out a big fat book about dinoseurs. When he got to the middle of the book a big, fat, man appeared in front of him. Then he heard somebody say.

'Think of a vish,

Just any old wish,

Just wish anything that

you want,

A bottle of rye.

A plum in your eye.

Vish. wish. anything that

you want."

Charlie thought for a minute.

"Hurry up kid!" the fat man said. "I wish that . . . "

"Well, come on!" he said again. "I'd like . . "

started Charlie. "Come on, come on! Hurry up!" " I wish
that I could be back in time, in the time of the
dinosaurs... For a week!" "Your wish is my command!"
Then the fat man disappeared. Charlie thought, "Well, I'm
going to need some food, and clothes and a toethbrush and
toethpaste. This is going to be fun!" Charlie went into
the kitchen first. He had three bananas, 4 oranges, 5
apples and 2 cans of soup in a bag when all of a sudden he
disappeared. Charlie respeared sitting on top of a tree in
a Jungle. "Where am I?" wondered Charlie. "In the time of
dinosaurs." said a voice. "But, but, I didn't seven have a
chance to pusk!" "You won't need that stuff anyways!"

Charlie cat in the tree for a little while. Then all of a sudden the tree began to shake and Charlie fell out. "THUMP!!" went Charlie as he landed. "I can't believe it..." thought Charlie. "I'm actually in the time of dinoscure." Charlie was really amased at this. Then he thought, "but where are all the dinoscure?" All of a sudden Charlie saw this big green monster in the distance. "A dinoscur!" thought Charlie. "I want to go and see if he will be my friend for this week." And Charlie was off!

Charlie was about 15 feet away from it when he saw the dinosaur licking it's lips. "W-Will you be my friend?" Charlie asked the dinosaur. The dinosaur shook his head up and down. "Good!" said Charlie. "I think I'll like you a lot." Charlie began to walk towards it. The dinosaur opened his mouth and popped Charlie in whole, without even chewing him. "Sure is dark in here." thought Charlie, "but I'm hungry!" Charlie pulled out 1 banana, and 1 orange. After he ate them he said, " I'm lonely, I want to go home!" and he started to cry. All of a sudden the big fat man from Charlie's story book popped into Charlie's velw. "You again . . . " said Charlie, "Take me home! I don't like it here any more!" "Are you sure?" asked the fat man. "Sure am!" replied Charlie. And so it was. Charlie went back home and locked up his book forever.

Michelie

Story 1 - Computer

MARY FINDS A FRIEND

One day Mary was out picking strawberries when she happened to notice a bear cub. Hary thought that he was sort of cute, so she went to see him. At first the cub seemed to be limping, so Hary decided to take him home so she could fix his leg.

Once Mary got home she took the bear cub up to her room. She snuck some food for the cub and some bandages for the cub's leg. After Mary had fixed up the cub's leg, she went to find some wood to build a cub house for her new pet.

Mary finished the cub house about 2 hours later. She put some blankets inside of it and brought out the cub.

"What should I call you?" thought Hary. She heard the cub grumble and then said, "What's that you say? You would like to be called Harvey?!" The cub grumbled some more and shock his head in a way that looked like he was saying yes.

Later on in the evening, Mary thought that she should ask her parents if she sould keep Marvey. After dinner Mary said that she had a surprise for her family. She brought down a big box with the name Marvey on it: She

PAGE 149
BLANK PAGE INSERTED

said.

"This is my surprise!" Hary set the box down on the table and pulled Harvey out. Hary's brothers gazed in amazement at the bear cub. Hary's mother said, "Where did you find this bear cub?"

"I found him in the forest, mother. He was all alone and was limping, so I decided to take him home so he would have a place to stay!" said Hary. "Please, please can I keep him? He's really nice and friendly! He wouldn't hurt me for the world!!" Mary's parents started to think. Her father replied.

"I'm corry honey. We just can't keep a bear in or around the house. It's too dangerous. What if he went out one day and came back with a bunch of his friends? Then what would we do?"

"Your father has a point there, Hary." chimed in Hary's mother.

"But Hom! Dad! He's so friendly!" Her mother and father thought about this for a little while.

Later on Hary's parents came up to her room to talk about the bear cub. They said that she could keep it over night, but would have to take it back in the morning. Hary agreed to this. The next morning Hary got up bright and early because she wanted to spend a little while with Maryey. She hussed and kissed him goodbye and then took

Harvey back to the woods.

"Bye, bye!" said Mary as she walked away. Hary never saw that sub again.

Michelie

Story 2 - Computer

SAH

Once upon a time there was a boy sailed Sam who loved to play jokes on people. But one day somebody decided to play a Joke on him.

Sam was playing on the monkey bars when he heard his name being called. He looked around, but there was nobody. Then he got off the monkey bars and started to walk around looking for the person who was calling his name.

Later on when Sam was at home, he heard his name being sailed again.

"What?" replied Sam, but nobedy answered. "I'm going to go and talk to my mether about this." said Sam. So he went inside the house. "Hem!" yelled Sam. "Where are you?" Sam looked around.

"You dear!" he heard his mother say. "I guess you want to talk to me, right?" questioned San's mether. Just as San was about to go upstairs to tell his mether the problem, she yelled down, "San't come up hency, I'm getting dressed." San ran dewestairs and out the door.

"Hever mind, Hem!" he yelled, "It's ekey. I just figured out how to solve my problem!"

Sam went back to the park and started playing on the playground. He heard his name being called again so he went to hide around the corner. Luckily Sam found the person who was calling him.

"SAM!" his name was being called again.

"What?" said Sam. "I guess that I finally found you out!" The person said that he was sorry and the two boys became good friends after that.

Michelle

Story 3 - Computer

Si ak

On the night before the first day of school, I told my mom that I was really excited. I asked her if it was normal to be excited over the first day of school, and of soarse she said yes. So I got all of my stuff ready in my bas for the 60th time. It was the night before the first day of school, but I wanted to go to school that night.

The next morning I woke up on my own, but about 2 hours ahead of time. I went up to my mom's room to wake her up for school. I said.

"When is the bus coming most? Did he forget where I

My man shock her head in a way that looked like no. Then she told me to go back to bed.

Hy mem walked through my bedresm door to wake me up.

"Time to get up honey!" she said. Then she pulled the severs off of me. She looked at me and told me that I looked protty. I said thanks.

I ate my breakfast, brushed my teeth, and then brushed my hair.

"I'm ready!" I yelled to my mem. get my coat on and

left.

Of coarse we didn't spend much time at echool, but our teacher told us about all the fun things we were going to do the next day. I went home really excited.

In the morning I wasn't feeling good so my mom kept me home. Then I thought,

"Now I'm going to miss out on all the fun."

Michelle

Story 4 - Computer

THE BIG ADVENTURE

"Help! Help!" yelled Ralph as he ran away from the three kids in grade eight. Ralph ran down the back alley, but he came to a dead end. He turned around to see if those big kids were still chasing him, and sure enough they were. Ralph was hoping that they wouldn't notice him, but they did. The boys caught him and threw him in a big sarbage bin. After they left Ralph got out and picked all the garbage off of himself.

Once Raiph got out of the alley the grade eight kids maid to him.

"Who said you sould get out yet?" so Raiph ran as fast as he sould. He went through a door where the kids souldn't find him.

"Get out of here!" said a voice, "All we sell are little restangular objects which kids like you aren't usually interested in." he said.

"Oh yeh!" replied Raiph, "Then why would I have about 105 books that I've read a thousand times?!"

"Ch. but those are probably semis backs as usual."
"Well, they're not! There's 'THE BLACK STALLIGH', 'THE

VIEW FROM THE CHERRY TREE' and a whole bunch of others like that!"

The man shock his head and continued reading his book.

"What's that you're reading?" asked Raiph.

"Have you ever felt when you read a book that it's really happening to you?" Ralph shook his head up and down. The man continued, "Well, this is that type of book. But you would never understand it!"

Then the phone rang. The man got up and answered it. When he came back the boy was gone, but so was his book.

"Oh no!" thought Ralph as he got to school. "I'm missing my math test!" He pecked in the room. All the kids had started their tests already, so Ralph went to the attic when no teachers were around.

There were spiders and spider webs all over the place. Raiph looked around the room and saw about 5 skulls piled up in one place. There was a dirty window along one wall. Raiph pulled a dirty mat out into the middle of the room, sat down and began to read.

THE DIS ADVENTURS. Raigh road. Raigh road and road the back until he got hungry. Then he pulled out his apple and started munching every. Raigh continued to road the back until he was almost done. He got up and went to look out the window. He spended it up and looked out. He gow

the streets below and thought, "I wish that I was in that story. It's sounds really exciting!" Ralph looked back at the book, closed the window and went back to read.

Raiph was just a couple of pages away from finishing the story when he relized it was very quiet in the school. He went out and looked around the school. It was raining and everybody had gone home. Raiph went back to the attic and tried to finish the book, but he was just too tired so he decided to so to sleep for a little while.

The next morning when Raiph wake up, it looked nice and sunny out. He ran up to the window and stared out for a little while. Then he went and sat down. He picked up the book and finished it. Raiph put the book back in his bag and ate a little bit. Then he looked out into the hallway to see if anybody was around, but nobody was. He checked in his classroom window and sew that kide were in there working. Raiph decided to return the book before soing home.

Michelle

Story 5 - Computer

The Recheaded Girl

One day when Susy went to echool everybody started laughing and teasing her for no reason at all. She ran into the washroom and started to cry.

At about 9:46 a.m., Susy went back into her classroom. She sat down and pulled out her math book, spened it up to the page assigned, then said,

"I don't know how to do this!" The whole class looked at her.

The teacher came over to Susy and explained what to do.

Later on as Suzy was walking home, two boys came up to her and started pulling on her red hair. She yelled at them and then ran away. Suzy felt embarassed.

In the morning she got a black felt and colored her hair black. When she went to eshcol, the kids treated her normally. But when she got home her mother tried to wash the black felt out. Luckly for Suzy it worked.

The next day when Susy's mether wasn't looking, Susy stole a wig. At school she put in on and although it was a little bit too big for her, she were it anyways.

During class when Suzy put her head down, her wig fell

off. A few kide looked at her, then some more kide looked at her, and then the whole class started laughing at her. Busy felt embarased and went to the washroom for a few minutes, when she returned everybody smiled at her.

After a couple of days the kids didn't tease her as much anymore because they finally got used to her red hair and they would also get in trouble if they did.

Michelle

Story 6 - Computer

Allen

One day Allen went into the kitchen to eat breakfast. He pulled out the Alphabets and poured them in his bowl.

"Take it easy, take it easy!" Yelled a voice from the box. Allen looked around, but nobody was there. All of a sudden a little man popped out of the box. "Hi there!" said the man. Allen said 'hi' back. "I'm the man from the Alphabets box, don't you see." The little man went up to Allen's bowl and said. "That spells elephant, purple and pink. And look, there's a lion, orange I think!" Then he sat on the edge of Allen's bowl and started to cry. "Look, look," he said "Look at that poor little fly. Everything I see and then touch always dies or gets hurt somehow." And then he hid his face and cried a little bit more.

The little man went to sleep in Allen's bed while Allen thought about his problem.

"A magic touch is all he needs." Then Allen dialed fun-words.

"Helle" said a voice, "Can I help you?"

"You!" replied Allen. Then the woman on the phone

said.

"Well, what do you want? Please tell me your problem."

Allen told the lady the whole story. Then she asked if Allen could bring the little man over to 'THE FUN POUNDATION' to see if she could help him out.

In about 15 minutes Allen and the man from the Alphabets box were at 'THE FUN POUNDATION'. The woman looked at him, made funny noises at him, and also tickled him. The lady asked to talk to Allen alone about the little man's problem and she told him.

"I'm afraid that your little friend has a big problem. I'm going to have to take him into the lab for a few days to study his habits. Then when I give you a call you can come and pick him up! If his problem is as big as I think it is, then he's in trouble!" Allen agreed to it, said goodbye to his new friend, and left.

A few days passed by and Allen got really worried about the little man when all of a sudden the phone rang.

"I'll get it mami" yelled Allen to his mother.
"Nelle!" he said

"Yes, Allen, is your mem heme? um. This is Mrs. Douglas."

"Just a minute, I'll get her!" Allen ran upstairs and told his man that Mrs. Douglas was on the phone. His

mom talked and talked and talked on the phone until Allen got worried that the lady from 'THE FUN POUNDATION' had tried to call him. He told his mother to get off the phone because he was expecting a phone call. She said goodbye and hung up. Allen sat by the phone all day until the lady called him.

Finally after he ate dinner, the phone rang. The lady told Allen that he could pick the little man up anytime, so Allen went right after he hung up.

Once Allen got into 'THE FUN FOUNDATION' building, he went straight to the lab. He walked in and there was Allen's friend. He was sitting on a bed in the room and as happy as sould be. Allen went up to him and hugged him. Then they started to talk. About ten minutes later the lady walked in the room. She took Allen into a different corner of the room and started talking to him.

"I studied him and I'm glad to say that nothing is really wrong with him, except for the fact that he isn't feeling so sure about himself. I have some special pills that he must take each day. It says on the bettle how many he's supposed to take and at what time. It doesn't matter if he runs out because this is all he'll need to take. If he ever needs anymere help, please some back to me."

Allen and the little man went back home and went to

pills. Then he would practise his magic on the dog, and eat. After the little man started taking the pills, he felt like a new person!

Meghan

Story 1 - Paper and Pencil

Barth Worm City

Once upon a time, in a far away land, there was a city for earth worms only.

The worms in this city were very excited, because a rock star called Wormy was coming to town.

This rock star was the coolest worm in the world. He wore black leather pants, a red silk shirt that was usually only done up half way, and his hair was usually greased down. When all the girls saw him (evan if not in person, Just a poster) they went wild! They also went wild when he went on stage to sing the song he dedicated to all the beautiful wormy girls in the world.

Pinally the day came. Wormy arrived at Barth Worm City. He was hoping that he could get there early early so he wouldn't have a croud of girls to get through. He arrived at \$100 in the morning. "Good!" he said, "How I won't have a croud of girls to get through!" But Wormy was urong. The girls had been lined up at the door waiting for Wormy scince 11:00 that night. "Go around to the back door. There may not be as many girls there!" Once again, Wormy was urong. There were just as many at

the back door as there had been at the front door. "Oh well. I'll just have to fight my way through!!" He made it through successfully, only having 18 worms kies him and all of htem touch him. Just as he entered the wormsciam, he noticed a very young and beautiful worm standing off to the corner and watching. Wormy knew he was in love. Love at first sight! He made himself a promise to get to know this worm.

After five hours of practicing for the show, he went out to find the girl of his dreams. She had blond hair that came 1/4 of the way down her body and curled towards the ends. She had a light purple bow in her hair with white polka-dots. She was wearing a blouse and pants that were the same shade of purple as the bow in her hair.

Just as Wormy was about to give up, he spotted her, in Worming Dales. 'Okay,' he said 'be sool, after all this girl'may not like you.' He tried to satch her eyes with out being too obvious. When she was coming out, Wormy said in his most worm attresting voice, 'Hi there.'

"You're Voray!" she said.

"You. Would you like to go out on a date with me?" he asked.

"I'd love to!!" she said.

*Good. I'll pick you up temorrow might at 7:30.

After a few more dates, Wormy popped the question, "Will you marry me?"

"I'd love to!!" said Kathy, the worm he loved. They married and lived happily ever after.

Heghan

Story 2 - Paper and Pencil

Babyeitting

"Casey! Jonnie!" called their mother. Alice. Both cirls came downstairs.

"Jonnie, could you babysit your cousins on Friday when we so out to susper with your aunt?"

"Oh mom I can't. I'm going out with Bobby that night. Haybe Camey could." she answered. Jonnie was 15 years old. Everytime a babyeitting job came along, she was always asked. To babyeit you have to be 12 years or older. Camey was only one year younger then Jonnie yet she never got to babyeit.

"All right. Will you bebysit them?" she asked. Casey had been dying to beby sit her cousins. Her aunt just had a beby 6 weeks ago. Her name was Angela. The next youngest was Joshua. He was 3 1/2 years old. He looks just like his father. Dark brown hair with big blue eyes. He is very loving always going around kissing people. His father died in a car crash when driving home from work, when his ount was three months prognant. The oldest of the kids are Tanya and Timmy. They are twins and both in grade two. Tanya is 7 minutes and 44 seconds older than Timmy and three inches tailer then Timmy who's 3 feet.

and went to put it in her mouth. Angela pushed it away with her hands and started kicking her feet.

"Oh Angela." sighed Casey. "Why won't you eat? Why can't you stop crying?" Just then Timmy walked in.

"Why don't you change her pampers." suggested Timmy. Casey groaned, long and loud. She picked up Angela and walked into her room. She set the baby onn the change table. Casey turned around to call Tanya and saw Joshua standing in the door watching. "Where does Hommy keep the pampers?"

"follow!" said Joshua. She followed Josh into the bathroom. Underneath the sink were the pampers.

"Thank you" Casey walked back into the baby's room.

Joshua followed her. Casey carefully undid the pamper.

She was relieved to see it wasn't anything muchy and gross. After that Casey laid down Angela and gave her the bottle. Casey watched to with the twins and Joshua while Angela sleet.

"So how was bebyeitting?" askeed Alice.

"It was okay the kids were really good."

*Okay, would you like to babyeit then again next week?" asked her mother.

Campy sighed. "Sure" she answered. All the thoughts of the evening ran through her head. What if next time Angela did something muchy and gross. Oh well!

Mechan

Story 1 - Computer

ROSE BUD

Rose Bud is a young flower that lives all alone in a forest. She is sad because she has no other flowers to talk to. Some times Rose Bud will make up pretend friends to play with.

One day in May, Rose Bud was playing with her pretend friends, when she heard the crunching of twigs and leaves. She wasn't sure what to do. In all the time Rose Bud had lived in the forest, she had never heard this sound before. Suddenly she saw a big pair of feet in front of her. At the sight of this, Rose Bud fainted!

When she mucke, Rose Bud found her beautiful stem soaked in a vase of water. "Oh no!" shrieked Rose Bud, "Samething has flower-napped me, and is trying to kill me!" Rose Bud began to weep getting her soft red petals covered in saity tear drops.

After a few days, she found herself turning brown. Her stem was starting to get all hard, her petals were beginning to fall off one by one, and she was dying. After a few more days, Rose Bud said, "The end is coming soon, I'm about to die any time now and no one is here to keep me company in my last few days of life."

That night when every one was a sleep, Rose Bud died.

No one had noticed her death. About three days after her death, the people who brought her from the forest, saw that she was dead, and threw her out.

As she hit the ground, something fell out of her pistil. It was seeds.

After about three weeks, little plants began to sprout out of the ground. In a few months, there were little flowers to take their mother's place, and they grow up to be happy because they always had each other to keep them company.

Heghan

Story 2 - Computer

Toby the Turtle

Once upon a time, there lived a turtle named Toby. He lived in a fish bowl owned by Luke. Once in a while, Luke would let his turtle out of the bowl to walk a round.

One day Luke took Toby outside to get some exercise. He put Toby on the lawn, then went inside for a glass of water. When he came back outside, Luke could not find Toby. "Toby, Toby!" Luke called. After subile, Luke got down on his hands and knees and felt around on the grass for his turtle. He searched the whole yard for Toby, but couldn't find him.

He went incide and began to cry. After a few hours. Luke's parents came home from the horse races.

"What's the matter Luke?" asked his mother.

"I lost my turtle, Toby." he said.

"Well, how did you do that?" asked his mother.

"Well, I took Toby outside to walk around in the grass, then I went inside to get a glass of water, and when I same back out, he was gone." said Luke. He started to sry again.

Luke and his mother went outside to look for Toby.
"I found him!!" yelled Luke after they had been looking

"I'd love to, mom!" said Casey.

Finally the day came.

"Okay you three," said Aunt May. "You be good for Casey."

"Yes, mom" they all answered. Aunt May started to tell Casey things to do about the baby. "She wakes up every 4 hours unless she's hungry. Her bottle is in the fridge and you can heat it up in the microwave for 3 minutes on low. Her food is in the pantry on the Srd. shelf. Today she is supposed to eat the peaches. Her bib is in the drawer with the tea towels. "I got to go. We'll be back at 9:00. By!" After her aunt left they watched tv for shile. Suddenly the baby started to cry. Casey walked into the baby's room, and picked her up. "Don't dry Angela." Casey said. Still holding her she walkedinto the kitchen and popped a bottle into the microwave. "I'll just set you down here" She tested the bottle. "Here you are Angela." Casey stuck the bottle in her mouth. She still didn't stop grying. Casey put her on Tanya's knee. "Can you hold her for a while?"

"On sure." she said. Casey walked into the kitchen. She took out a jar with the word peaches on it. "Tanya can you bring in Angela?"

In walked Tanya with the crying baby. "Just set her down in the highshair." She took a bit of much on the speen

for a while. "He must have been hiding under the shed." He took him inside and gave him some food.

Heghan

Story 3 - Computer

Hystery of the House Eyes part 2

Mary Jane started to slow down. I could tell she was getting experemly tired. I jumped off the bike and started to run a long side of her. "They'er coming closer Mary Jane" I yelled. She jumped off the bike. We both started to run. We headed for the high way, (that way if they tried to pick us up people would see and here us call for help, it was also shorter). We made it home all out of breath.

"Hom!" gasped Mary Jane, "3 men, they are at Uncle Jim's house..."

I gut her off "they killed him" I cried. tears started pouring out of my eyes.

"They'er coming here to get us" Hary Jane said "We half look our doors". "Girls, you

sure are acting strange. Uncle Jim is not dead. I'll take the car over there and show you.". She thumped out of the house, slaming the door behind her. "Hom." yelled Mary Jane

"Hary Jane, I know what you are thinking, and we

dan't go out there. They'er after us, remember?"

"I know, nut if they get mom...". She stoped talking and taers started to roll down her cheaks.I'd never seen her like this. Her long blond hair all messed, (the only time her hair we like this was in the morning after she got up), her beutiful complection(her face) all scrached up. Poor Hary Jane.

I wish I could do something to cheer her up. She looks so glum. "Mary Jane, whats the matter?" I asked.

"Oh nothing. I'm just a bit nervus about mom going over to Ungle Jim's house alone."

"Don't wory, she'll be all right, I can garente it."
I said.

"I hope so." said Hary Jane. Just as she said that, mom same threw grying. "Oh mom, whats the matter?" asked Hary Jane.

"It's Uncle Jim, you were right, he is dead!" mom broke out in to tears. Hary Jane and I sat down beside her. Poor mom. Uncle Jim was her only brother. They were really close for being brother and sister.

"It's okay mom." Hary Jane said

"No it's not. That would be like me saying, 'It's okay' if your sister died."

Out of the corner of my eye, I saw some thing that

looked familer, the blue van, the men inside they are here to kill us. I ran towards the door, and put my hand up on the lach to lock it. "Mary Jane, they are here." Hom ran to help us.

"Janis, so lock the back door!" my mother comaned me.

I ran into the kitchen. Beside the stove was the back
door. I locked it. Just then the kettle wisteled. I had
a perfict idea. I poured the hot water into 2 tea cups.

Then I went outside, with the hot water. When I was
behind the 3 men who were trying to get in, I poured the
hot water on them, then ran into the house locking the
door behind me. "We're safe for now!" I called to mom and
Mary Jane. As soon as they heard that, Mary Jane
ran to lock the door which mom was holding shut.

"What did you do?" asked mom.

"I poured the hot water that was on the stove into 2 tea cups, took them outside, and dumped the water on them."

"Oh Janis!" said Mary Jane, "You are so brave!"

"Thank you, but we can't discuss that. The men will be back so we have to get out of here quick!"

"We could go to your grandma's house." suggested mom "That's a great idea mom." said Mary Jane.

"If we are going to grandma's then we brtter leave now." I said. We got into the car and off we went to grandas's.

When we got there, mom asked, "Do nyou mind if we spend a few days here?"

"Not at all." Grandma said in her old ladys voice.

We went up to the rooms we usually stay in when we stay at her house. "Hom, why didn't you tell her why we are here?" I asked. I knew what the answer would be, but I felt like asking.

"I didn't want to worry her. She is an old lady, she couldn't take if we told her about Uncle Jim's death, and the men coming after us." she said.

"Ch." I said.

After 24 games of solitary, Grandma called us down for supper. "I'll be down in a seconed" I called to her. I finished picking up the cards, and went down stairs. When I got down I saw Grandma crying endlessly. I went over to Hary Jane and wispered, "Hom told her didn't she?"

"Yea. She's takl: It pretty bad too. Poor woman." said Hary Jane in a low wisper. Supper was pretty quiet that night, exept for the little sounds Grandma was making as she cried. After Grandma went to bed, mom asked me a few questions. "Honey, about how tall were the men?"

"I don't know, about as tall as you." I answered.

"Did you see their hair seler?" she asked again.

"Mom. they were wearing "ki masks." I said. "I don't feel like answering any questions to night." I said.

While I was a sleep. I dreamt the men had killed all of us. I woke up screaming. How and Hary Jane had come running into the room. "They are going to kill us." I said.

The next morrning mom called the police. They sent someone over to keep an eye on us. Sort of like a body gard. The body gard said we'd have to come home, back to our house, because that was the only way they could catch the killers.

That night after every one was a sleep. (exept the person guarding) I heard the front door creak open. I shot strait up in bed and began to panic. The sweat was just pouring out. I ran into my mom's room. "Homi" I said in a low scream. "The killer, it's here in our house!"

She shot erraight up in bed just like I did when I heard the front door open. "Don't panic!" she said quivering. "Quick, go get your eister." I ran into Hary Jane's room. She was sitting staight up in bed.

"The killer is here she said in a low wisper."
"I know. Hom said to some get you."

Defore we even made it back to mom's room the lights

went on downstairs, and we heard our guard yell, "PREEZ!!" and saw him point a gun at the men we saw in the truck. I'm not sure what Mary Jane was feeling, probably puzzled, I thought because she had a puzzled look on her face. Suddenly we heard a buch of sirens, and saw the men being put into police cars. I turned around to go get mom when I saw her standing there watching in awe.

The next morning we got a phone call. "Hello, is you mom home?" asked the vocie when I answered

"Yes, will you just hold on a minute?" I ran to get

"Hello," she answered. She talk an awful long time. When she came back downstairs she said, "That was the police, they told me that the men would be fined \$600.00 and would be put in Jail for 3 years.

"YAHOOOOO!!!!" yelled Mary Jane and I togeather.

I went upstairs into my bedroom to lay down and relax. Suddenly I heards sound. Ot sounded like my drawer fell out of it's place. I turned around quickly, there was no one there. "Mary Jane!" I called, "I think we have another mystery!!"

Sarah

Story 1 - Paper and Pencil

How Earth got its Color

In the Amazena in 0001, l'anabush was playing with his friends of the forest. The forest was a dual color.

Nanabush thought for awhile and said, "I shall get some paints." The next day Nanabush walked one mile to the store. He bought red, orange, yellow, green, blue, brown, black, white and purple. He paid the man with furs.

Manbuch went back to the forest and called all the greatures. He painted the robins, brown and red. He painted the sky blue the meadows green the trees brown, and all the other wonderful things. Animals were more happier. And that's how the earth got its color.

Sarah

Story 2 - Paper and Pencil

Return of the Cube

Part 2

The cubs were getting big now, they were four months old. Rebecca and Fighter were starting too teach them murvival.

The first thing they had to teach them was too catch wild animals. First Rebecca and Fighter had to show them how to sneak up on their prey. When to chase their prey and how to spring on the animal. They learned very quickly. On the II day Gerard cought a rabbit. He was proud of his catch.

After each cub cought a rabbit, Pighter and Rebecca. then taught them who was their enemy.

heared a whistling sound coming from some bushed he was a curious cub. He quickly ran in. He let out a cry and ran cut of the bushes again and ran to his father. Pighter said, "I will not protect you. You are seared of just a little bird. I think you a coward." Poor Gerard, he thought, he never did like me." Then his father came over and said, "son your's not a coward but I didn't think you

should have run away from a little bird." "I'm sorry, father."

The next day there was a fawn sneaking by. Shenan, Gerard and Hollywood Gold quickly leaped on it. After a few minutes the fawn was dead and their parents were proud of their children.

Then one day the cube were playing by a campelte where people were camping. What will happen? Will they run home quickly to their parents? Find out in part three.

Sarah

Story 1 - Computer

The Problem

In the town of Cooking Lake, a boy named Toby, was busy doing his spring cleaning. "I don't want that," he yelled, throwing his old ragged blanket into the Junk box. After a little bit of cursing and losing his temper, he was done.

As he was taking his garbage out to the garage, his best friend. Tiny, came over to ask him if he wanted to play ball.

"Sorry I can't. I have to do some homework."

"Just come for a little while."

"No, I can't I said." They started to argue.

Very soon they were punching and swearing at each other.

When Toby's mother, Hrs. Brown came out, the boy's quickly stopped fighting. Then Hrs. Brown said, " I am ashamed of you both."

Toby answered quickly, "I told Tiny, I could not play with him because I have homework to do."

Tiny started to walk home dragging his bat and ball with him. He acreamed very harshly, "I dun't need a clumb friend. like you!" "You're so dust in math that you look over my shoulder and sheat!"

"Shut up, you are Just a nerd." yelled Toby.

"That's enough boys," yelled fire. Brown.

That night at supper, Mr. Brown asked, "how did your day go Toby?"

"Not very well. I had a fight with Tiny."

"Well, don't you think you should make up?"

"It was not my fault. I told him I had to do my homework."

Just then the phone rang. Toby ran to get it. It was Tiny. " Hi Toby, I was Just wondering if you want to play ball."

"Sure I will," replied Toby.

In the end it was back to normal until there was another problem.

Sarah

Story 2 - Computer

The Alley Cat

Have you ever stopped to look into an alley where date scrounge for food? Well, I sure have. I will tell you a story about an alley dat who was so weak that he almost died. But with lots of love I saved him. He's allve to this very day.

One day in the month of May, I was walking down the back alley of our apartment block. I enjoyed my walks in the alley because it was very peaceful. That day was different, though. I heard a sudden cry coming from some old garbage cans. Curious, I ran to see what was crying.

When I got there, I saw a little cat. He was very weak. I picked him up and carried him to the apartment. I left him by a pipe and ran to get a blanket. There were no pets allowed in the building. I had to sneak him in.

When I finally got him into the apartment, the snobblest girl in the building came by and asked, "What do you have there?"

"Nothing you would be interested in", I replied nastily.

I quickly walked on and went into our apartment. No one
was there so I was safe for a while. I went to my piggy
bank and took out seven deliars. I left the cat in my room

in my closet.

I bought a bag of cat food at the store and rushed home. Running into my room, I rushed to my closet. I poured a little dish of food for the cat. He gobbled it up fast.

A week went by and the cat was getting strong. I decided to call him Fluffy. When he was strong again, I asked my friend if she wanted a cat. She asked her parents if she could keep Fluffy. They agreed. That day I brought him over. Now he was better and he had a home. He was very happy there.

Sarah

Story 3 - Computer

Rusty

good one, till mom saw my dog Rusty lying on the ground half dead. Hom ran to get her. She put her on a hoard sway from the ground. When I was eating breakfast, I was almost in tears. Hom said, "I don't want Rusty to suffer any more."

"I know" replied ded.

When we went outside, she had blood coming out of her nose. Within two minutes she was dead. We buried her in our garden.

Two more days went by and we got a new dog. He's a good dog. but I still miss Rusty.

Sarah

Story 4 - Computer

The Clue

It was a beautiful night. We all sat around the camp fire. The flames of the fire were bright with orange and yellow hues. Everyone was telling Jokes. I sat listening to the Loon's peaceful singing. It was like I was in a trance. I quickly fell asleep.

After what seemed four hours, I heard Patti in the background. She was saying something about a wild horse. I opened my eyes but there was no one in sight. I felt a shiver coming down my spine. Everybody must have gone back to the cabins. I quickly jumped up and ran to the cabins. The only person who was there was Patti.

"Did you call my name?" I asked in a suspicious voice.

[&]quot;No," she replied.

[&]quot;Then why were you calling me?"

[&]quot;I ween't."

[&]quot;Where are the others?" I asked.

[&]quot;I thought they were with you."

[&]quot;They were, but when I wake up they were not there."

[&]quot;Semething is fishy", I said.

Chapter one

The warning

We started hunting for clues. Patti and I make very good detectives. I walked to the beach to see if there was any sign of the others. Just as I was turning to get back to Patti, someone tossed me to the ground. A piece of paper flue out of his hand. I quickly got to my feet.

I ran back to the cabins. The note had a strange message on it. The message read, CAVE W+h WILL LEAD YOU TO YOUR FRIENDS. "What does it mean? This message is all cramped up in my memory. I know it means something. But I don't know what."

CHAPTER 2

The Theory

"In my dream, you told me comething about the Wild Horses. The other day while I was hiking I saw a outline of a horse drawn on a piece of resk," I said.

"I know, it might be a clue to the riddle. If you take me to the rock, we might figure out the message," exclaimed Patti.

"I think I still remember where it is."

After what seemed like an hour, we arrived there. The drawing was where I found it. Just then we heard a remailing sound in the distance. It was coming from a cave, not far from where we were standing.

"Its like a volcance," we both whispered.

Chapter 3

Safe

We ran over to the cave. We saw a light shining through an opening, at the bottom. Curious, we walked over to the pile of rock. We heard an echo in the cave. We shivered with fright. We started to throw the rocks off the rock pile. It took a while. We made a hole big enough to fit into. I crawled in. Everyone that had been missing was huddled together. They quickly ran to meet me. Just as we crawled out of the cave, we saw Pattifall on a sharp stone. I ran over to her and said, "What happened?"

"I don't know."

Two of the strong boys sarried her home. After we got back to the sabins we told them what happened. I said to Patti, "We actually solved a case.

Joan

Story 1 - Paper and Pencil

How Cats Eyes Glow

Long, long ago in a far away land called Silver Shade, there lived many different knids of cats. Each cat had a name to fit the color, shape and size of them. Why don't I introduce the characters in the story. The first character I will introduce is Freddy Pathead, the cat. The reason he is called that is because he has a fathead and a small lean body. The next character I'll introduce is Nanaboso. Nanaboso helps animals and other living things including people. Last but not lease is Sam the slick cat. Sam is the coldest and the most slickest cat of all. Any way let's get to the story.

One day Freddy the fathead was taking a walk around the park, when he meet Sam. Freddy and Sam are really good friends since they were kittens. While they were walking they meet Hanabush. They asked Hanabush if he would like to join them. So off walked the two cats and Hanabush. While they were walking Hanabush asked his two friends if they would act out the story Hanabush was going to tell some children. About an hour later it started to get dark. So Freddy and Sam and Hanabush went to where

Nanabush lives. When they got there the shildren were waiting for their bedtime story. The children lived quite near to where Nanabush lives. This time Nanabush was going to tell the "How Cats Eyes Glow".

One day there was two cats that were very unusual. They would always sleep in the day and hunt for their prey at night. But the strange thing about it was that Sam and Freddy would stare at the moon because they needed light to guide their way. The one night Sam and Freddy didn't have to stare at the moon because their eyes started to glow. Soon the news spread and all of the cats in the world stare at the moon. And that's How Cats Eyes Glow.

The

End

"Now off you go to bed!" Manabush told them. So the children ran home and went to bed.

Joan

Story 2 - Paper and Pencil

(Untitled Story)

One day Emily Mayer was sitting on her bed reading a Sweet Valley High Love Letters #17 book when the phone rang. Karen, Emily's stepmother answered the phone. "Emily dear Dan wants you on the phone." Emily picked up the extension in her bedroom.

"HI Dan what yah doin?" Bmily asked.

"Nothing really, but I'm wondering if you would like to go to the school dance on Saturday night with me?"

"Sure I'd love to!" Emily answered back. This was Emily's big chance to get to know Dan better.

"Well, I've got to go now my mother wants me to clean my room, Bye." Dan said hanging up the phone. Emily hung up the extension and continued reading.

The next day was Friday. In the Mayer's house Emily was running around trying to find something to wear to the dance Saturday. She wanted to impress Dan. He was like a dream that was coming true for her.

Soon it was Saturday afternoon. Buily wake up and heard the birds singing. Her dog ran to greet her. He is a weiner dog. He has floppy brown ears and a long flat

motted brown and white body.

Shopping mail to get a new outfit for Dan. They lived about a block away. Pirst she went into Antels. There was nothing for her. The second place she stopped at was Reitmans. Smily found this beautiful dress. It was baby blue with puffy lace sleeves. Knowing the dress was made for her she bought it. After she got the shoes, earrings and other fine Jewelry, she started towards home.

Everything was going to be perfect.

It was 4:00 in the afternoon when she got home she was so excited about the dance an even more importantly. Dan. Soon 6:00 came and Emily started to get ready. First she put on the fabulous dress, then fixed her light golden brown hair. 'Emily dear, would you come down here for a moment? That is if you have the time." 'Sure I do!' Emily said. So she walked down the stairs to where Karen was standing.

"Bmily, I would like you to bebyelt Karrie for me.
You know bebyeltters now-a-days. They fall a sleep on you and then to top it off they want twenty bucks.

Your dad and I won't be gone long, bye" Earen said rushing off. Emily was about to burst into tears. Of all the nights her stepmether had to pick the night when Emily had plans.

Ding Dong went the bell.

"Oh great, Dan's here already." Emily thought. She walked up to the door and opened it. Sure enough it was Dan. "Come on in, there is a slight problem. I have to baby sit until my mom and dad come back home. Dan was so understanding.

He watched the baby while Emily was getting ready.

About an hour and a half later Karen and Mike walked in.

Karen asked a few questions about Karrie and then let

Emily and Dan go to the dance.

Emily and Dan got to know each other really well. They ended up having the best time of their lives.

Joan

Story 1 - Computer

TEENAGER'S KNEES

Teenager's knees is an injury that most sport participants get. When you bend your knee, your knee cap and a bone behind the knee cap grinds together, causing pain.

Some treatments will help reduce the swelling and improve your condition. The best treatment for teenager's knees is a lot of rest. Ice will help bring down the swelling and reduce the pain. Some doctors may prescribe medication if necessary. When you have this problem you will have to make some adjustments. Your doctor may prescribe a brace or something to hold your knees in place.

When you're in this position there has to be little pressure on the knees. The best activities are the ones that limit the knee to range between 185 degrees and 180 degrees. Some good sports are: swiming, slow - Jossing or walking, skating, and cross-country skiing.

To improve your condition some people have to strengthen their quadricop muscle. The dostor will recommend cortain exercises. There are about five

different stretches.

Take it from me, it hurte!!

Joan

Story 2 - Computer

GOODBYE. SWEET VALLEY HIGH

Down in Sweet Valley High Elizabeth and Heather were sitting on the patio sipping coke slurples and having a friendly conversation, when the phone rang. Elizabeth picked up the new extension, that her father put in.

- " Hello" Elizabeth said as she put down her slurple.
- "Hi,my name is Natasha Haloney, the real estate manager. I hear that you want to sell your house. I might have possible buyers. Will you be home today?"

"Sure I will! What time will you be bringing them over here?" Elizabeth questioned.

"About 1:00 o'clock. De ready!"

After Elizabeth hung up the phone, she glanced at the clock.

"Twelve-a-clock!! We've only got an hour to clean the whole house," Elisabeth said, screaming at Heather (Elisabeth's friend). After Elisabeth finished screaming at Heather, they started to clean the house. It didn't take them very long to clean up because there were two of them.

About an hour later the real estate manager was at the door.

"As I told you, I was going to bring a very nice couple to see the house." Elizabeth welcomed Hr. and Hrs. Fowler to the new home. Natasha showed Hr. and Hrs. Fowler around the house. At 2:00 o'clock the Fowlers and the real estate manager left.

Heather didn't want her best friend to move, but she knew Elizabeth had no choice. Heather was Elizabeth's friend since kindergarden.

About an hour later Jessica and her mother came walking in with arm loads of groceries, and bags which contained clothes for the twins (Jessica & Elizabeth). Elizabeth told her mother about the real estate manager and the Fowlers.

About a week later the family went to find out what school the twins were going to. The school was a block away from the Los Villas Estates.

Jessica and Elisabeth had mixed feelings about moving. They were glad that they were moving because they would have bigger bedrooms and a lot more space, but the twing would have to make new friends and adjust to their new school.

About a week later Hr. and Hrs. Povier bought the house. The Hipps (the twin's last name) got a one week notice on when they should move.

The very next day the Hipps started to get ready to

move. Elizabeth's friend. Heather didn't know when they were moving.

One Saturday Heather came over when the Hipps were packing their clothes. Heather asked Elizabeth if she would like to go to a movie, but Elizabeth turned her down. Heather started to think that her best friend didn't want her around anymore. So Heather tried to avoid her but it didn't seem to work!

A couple of days later Elizabeth asked Heather if she would like to come over and see her new house. Heather agreed for only one reason -- to talk to Elizabeth about the problem. It was only Heather's imagination. Elizabeth and Heather were best friends again.